Richard M. Cieri Ray C. Schrock Stephen E. Hessler KIRKLAND & ELLIS LLP 601 Lexington Avenue New York, New York 10022 Telephone: (212) 446-4800 Facsimile: (212) 446-4900

Counsel for Ally Financial Inc. and Ally Bank

UNITED STATES BANKRUPTCY COURT SOUTHERN DISTRICT OF NEW YORK

In re:)) Chapter 11
RESIDENTIAL CAPITAL, LLC, et al.) Case No. 12-12020 (MG)
Debtors.) Jointly Administered)

NOTICE OF FILING OF LETTER AGREEMENT BETWEEN ALLY FINANCIAL INC., OCWEN LOAN SERVICING, LLC AND WALTER INVESTMENT MANAGEMENT CORP.

PLEASE TAKE NOTICE that, for informational purposes only, Ally Financial Inc. ("AFI") hereby files the letter agreement by and among AFI, Ocwen Loan Servicing, LLC, and Walter Investment Management Corp., dated November 19, 2012 and attached hereto as

Exhibit A.

New York, New York Dated: November 19, 2012 /s/ Ray C. Schrock

Richard M. Cieri Ray C. Schrock Stephen E. Hessler KIRKLAND & ELLIS LLP 601 Lexington Avenue New York, New York 10022

Telephone: (212) 446-4800 Facsimile: (212) 446-4900

Counsel for Ally Financial Inc. and Ally Bank



K&E 24430626

EXHIBIT A

Letter Agreement

November 19, 2012

Ocwen Loan Servicing, LLC 1661 Worthington Road, Suite 100 West Palm Beach, FL 33409

Walter Investment Management Corp. 3000 Bayport Drive, Suite 1100 Tampa, FL 33607

Re: AFI/Ocwen/Walter Transition Services Agreement

To Each of the Signatories Hereto:

Execute the TSA by December 21, 2012. Ocwen Loan Servicing, LLC ("Ocwen"), Walter Investment Management Corp. ("Walter"), and Ally Financial Inc. ("AFI") hereby agree to negotiate in good faith to complete a transition services agreement that will be reasonably consistent with the transition services agreement attached as Exhibit 1 hereto (the "TSA"), and to enter into such transition services agreement no later than December 21, 2012 (the "TSA Deadline Date").

The Executed TSA Shall Include SOWs for AFI Services to be Provided to Ocwen/Walter. Ocwen, Walter, and AFI hereby agree to negotiate in good faith and to complete the statements of work for the provision of services by AFI to Ocwen and Walter to be attached as schedules to the TSA by the TSA Deadline Date, which shall be reasonably consistent with the statements of work for "Parent Services" that were entered into in connection with that certain Shared Services Agreement by and between AFI and Residential Capital, LLC, dated as of May 16, 2012 [ECF No. 387, Ex. A] (the "SSA"), unless such other terms are mutually agreed to among AFI, Ocwen, and Walter.

The Executed TSA Shall Include SOWs for Ocwen/Walter Services to be Provided to AFI. Ocwen, Walter, and AFI hereby agree to negotiate in good faith and to complete the statements of work for the provision of services by Ocwen and Walter to AFI to be attached as schedules to the TSA by the TSA Deadline Date, which shall be reasonably consistent with the statements of work provided to Ocwen and Walter to date for services to be provided to AFI.

<u>The Executed TSA Shall Include Pricing Methodology</u>. Ocwen, Walter, and AFI hereby agree to negotiate in good faith and to complete the pricing methodology for the TSA by the TSA Deadline Date, which shall be reasonably consistent with the pricing methodology of the SSA, unless such other terms are mutually agreed to among AFI, Ocwen, and Walter.

<u>The TSA Deadline Date May be Extended by Mutual Consent</u>. The TSA Deadline Date may be extended by written agreement of all parties to this letter agreement.

Execution Version

<u>Counterparts</u>. This letter agreement may be executed in counterparts, by either an original signature or signature transmitted by facsimile transmission, other electronic copy, or other similar process and each copy so executed shall be deemed to be an original and all copies so executed shall constitute one and the same agreement.

[Signature Pages Follow]

Acknowledged and Agreed as of the date first written above:

ALLY FINANCIAL INC.

By: <u>/s/ James G. Mackey</u> Name: James G. Mackey

Title: Chief Financial Officer

Acknowledged and Agreed as of the date first written above:

OCWEN LOAN SERVICING, LLC

By:_/s/Ronald M. Faris_

Name: Ronald M. Faris

Title: Chief Executive Officer

Acknowledged and Agreed as of the date first written above:

WALTER INVESTMENT MANAGEMENT CORP.

By:_/s/ Stuart D. Boyd_

Name: Stuart D. Boyd

Title: Vice President, General Counsel and

Secretary

EXHIBIT 1

Transition Services Agreement

AFI/MB COMMENTS 11-165-2012 TO CLIFFORD CHANCE COMMENTS OF 11-14-2012

Transition Services Agreement

by and between

Ally Financial Inc.
and
Ocwen Loan Servicing, LLC {[[and Walter Investment Management Corp-].]]}

Effective as of ____, 2012

AMECURRENT 703734421.6 30-Oct-12 10:13 AMECURRENT 703734421.10 16-Nov-12 14:56

AFI, Ocwen and Walter to discuss, for each Buyer Service, whether Ocwen or Walter, or both, will have control of the platforms, systems, personnel or other resources used to provide such Service. Where one party has direct or indirect control of the resources used to provide such Service, that party will be responsible for that Service. Where both buyers have control over resources used for such Services, each will be responsible for such Service. Based on this allocation of responsibility, the parties will determine whether there will be a single TSA or multiple TSAs between or among AFI, Ocwen and Walter. Ocwen and Walter will enter into separate TSAs.

AFI/MB COMMENTS 11-165-2012 TO CLIFFORD CHANCE COMMENTS OF 11-14-2012

[**To be updated]

<i>1</i> .	DEFINITIONS.	<i>1</i>
1.1 2.	Definitions. TERM AND RENEWAL	
2.1 2.2 3.	Initial Applicable Term. Extensions of Applicable Term. SERVICES.	4
3.1 3.2 3.3 3.4 3.5 3.6 4.	Services	677
4.1 4.2 5.	Affiliates and Subcontractors	9
5.1 5.2 5.3 5.4 5.5 5.6 5.7 5.8 6.	Relationship Managers. Governance Model. Reports. Regulatory Review. Books and Records; Audit. Books and Records; Audit. Informal Dispute Resolution Procedures. Continued Performance. FACILITIES.	910101010
6.1 6.2 6.3 7.	Use of Recipient Facilities. Supplier Facilities and Systems. Physical Security for Facilities. INTELLECTUAL PROPERTY AND PROPRIETARY RIGHTS	12 12
7.1 7.2 7.3 7.4 8.	Ownership of Intellectual Property. Development of Intellectual Property. Limited License to Use Supplier Work Processes and Software. No Implied Licenses. RECIPIENT DATA.	12 13 13
8.1 8.2 8.3 9.	Definition. Ownership. Data Security. CONFIDENTIALITY.	13 14
9.1 9.2	Obligations Excluded Information	

AFI/MB COMMENTS 11-165-2012 TO

CLIFFORD CHANCE COMMENTS OF 11-14-2012 9.3 9.4 9.5 9.6 9.7 9.8 10. 10.1 10.2 10.3 REPRESENTATIONS AND WARRANTIES.17 *11*. 11.1 11.2 11.3 11.4 Disclaimer. 17 11.5 12. 12.1 Insurance. 17 12.2 Risk of Loss. INDEMNIFICATION AND LIMITATIONS ON LIABILITY......18 *13*. Indemnification. 18 13.1 13.2 13.3 Indemnification and Limitations on Liability Relating to Negligence and Strict Liability.22 13.4 13.5 Insurance. 23 14. TERMINATION.23 14.1 Termination Following Sale. 24 14.2 14.3 14.4 *15*. GENERAL......26 15.1 15.2 15.3 Counterparts. 27 15.4 15.5 Force Majeure......27 15.6 Further Assurances. 28 15.7 15.8 Headings. 28 15.9 Independent Contractors. 28

AFI/MB COMMENTS 11-1<u>6</u>5-2012 TO CLIFFORD CHANCE COMMENTS OF 11-14-2012 15.11 Public Announcements. 29 15.12 Amendments and Waivers. 29 15.13 Severability. 30 15.14 No Third Party Beneficiaries. 30 15.15 Order of Precedence. 30

SCHEDULES

Schedule A-1	AFI Services ²
Schedule A-2	Buyer Services
Schedule A-3	AFI Provided Migration Assistance Services
Schedule A-4	Buyer Provided Migration Assistance Services
Schedule B	Governance Model
Schedule C	Pricing Methodology
Schedule C-1	Pricing for AFI Services
Schedule C-2	Pricing for Buyer Services
Schedule D-1	List of Buyer and Supported Facilities
Schedule D-2	List of AFI Recipients and Supported Facilities
Schedule E	Form of Supplement
Schedule F-1	List of Pre-Approved AFI Subcontractors
Schedule F-2	List of Pre-Approved Buyer Subcontractors

Schedule A-1 will (a) include ITG Functional Resource Area (SOW No. AA – BB) / Freference the applicable ResCap Service SOWs under the SSAI, each with an Applicable Term that expires on June 1, 2013, and (b) provide that, on June 1, 2013, the Buyer Services in SOW No. AA-BB will be replaced by SOW AAA-BBB / Freference the applicable Buyer Service SOWs based on the corresponding SSA SOWs and to be agreed upon by AFI and Ocwen or Walter (as applicable) and the Applicable Term for Buyer Services under SOWs AAA-BBB shall commence on June 1, 2013.

² Schedule A-2 will provide that (a) the Applicable Term for AFI Services within the ITG Functional Resource Area (SOW No. XX − YY) [←reference the applicable Parent Service SOWs under the SSA] which will expire on June 1, 2013, and (b) on June 1, 2013, SOW No. XX-YY will be replaced by SOW XXX-YYY [←reference the applicable AFI Service SOWs provided by AFI to Walter and Ocwen] and the Applicable Term for AFI Services under SOWs XXX − YYY shall commence on June 1, 2013.

TRANSITION SERVICES AGREEMENT

This Transition Services Agreement (this "Agreement") is entered into and effective as of ____, 2013 (the "Effective Date") by and among Ocwen Loan Servicing, LLC ("Ocwen" [or "Buyer"]) [and Walter Investment Management Corp. ("Walter"; collectively with Ocwen, "Buyer")] and Ally Financial Inc., a Delaware Corporation ("AFI").

WHEREAS, Residential Capital, LLC, a Delaware limited liability company ("ResCap") and AFI (the indirect parent entity of ResCap) have entered into that certain Shared Services Agreement effective as of May 15, 2012 (the "SSA"), pursuant to which AFI provided to ResCap Parent Services (as such term is defined in the SSA), and ResCap provided to AFI Reverse Services (as such term is defined in the SSA);

WHEREAS, Ocwen, Walter, ResCap, and various affiliates of ResCap have entered into that certain Asset Purchase Agreement, dated as of November 2, 2012 (the "APA"), pursuant to which Ocwen and Walter agreed to purchase certain assets of ResCap, including assets used by ResCap to provide Reverse Services under the SSA; and

WHEREAS, in connection with the transactions contemplated by the APA, Buyer desires AFI to provide certain transition services, and AFI desires Buyer to provide certain transition services, upon the terms and conditions set forth in this Agreement.

NOW THEREFORE, in consideration of the mutual promises and covenants contained herein, and for other good and valid consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties hereby agree as follows.

1. **DEFINITIONS.**

- 1.1 <u>Definitions</u>. As used in this Agreement, the following terms have the following meanings:
- "Additional Services" has the meaning set forth in Section 3.2.
- "Acquired ResCap Business" means a business (whether consisting of a legal entity, a group of assets or a platform of assets and personnel) that is acquired by Ocwen [or Walter] pursuant to the APA.
- "Affiliate" means (i) with respect to AFI, (a) those Persons with respect to which AFI directly or indirectly owns or controls more than 50% of the voting equity securities of such Persons, (b) those Persons that directly or indirectly own or control more than 50% of the voting equity securities of AFI, and (c) as to the Persons identified in the preceding clauses (a) and (b), those Persons that directly or indirectly own or control more than 50% of the voting equity securities of such identified Persons, and (ii) with respect to Ocwen [or Walter], those Persons with respect to which Ocwen [or Walter] directly or indirectly owns or controls more than 50% of the voting equity securities of such Persons.
- "AFI Services" has the meaning set forth in Section 3.1(a).

"Applicable Term" has the meaning set forth in Section 2.2.

"Assignment of Leasehold Interest" has the meaning set forth in Section 15.17(a)

- "Buyer Services" has the meaning set forth in Section 3.1(a).
- "Charges" has the meaning set forth in Schedule C.
- "Claim" has the meaning set forth in Section 13.1(d)(1).
- "Claim Notice" has the meaning set forth in Section 13.2(a).
- "Confidential Information" has the meaning set forth in Section 9.1.
- "Customized Services" has the meaning set forth in Section 3.3(a).
- "Damages" has the meaning set forth in Section 13.1(a).

<u>"Eden Prairie Owned Real Property"</u> means the real property located at 6875 Shady Oak Road, Hennepin County, Minnesota.]

- "Equipment" means computer and telecommunications equipment (without regard to the entity owning or leasing such equipment) including: (a) servers, personal computers, and associated attachments, accessories, peripheral devices, printers, cabling and other equipment; and (b) private branch exchanges, multiplexors, modems, CSUs/DSUs, hubs, bridges, routers, switches and other telecommunications equipment.
- "Extended Applicable Term" has the meaning set forth in Section 2.2.
- "Force Majeure Event" has the meaning set forth in Section 15.5(a).
- **"Functional Service Areas**" means the categories of AFI Services or Buyer Services that are set forth under <u>Schedule A-1</u> or <u>Schedule A-2</u>, as applicable.
- "Governmental Authority" means any government or political subdivision, board, commission or other instrumentality thereof, whether federal, state, local or foreign.
- "Include" and its derivatives means including without limitation. This term is as defined whether or not capitalized in this Agreement.
- "Indemnified Parties" has the meaning set forth Section 13.1(d).
- "Indemnifying Party" has the meaning set forth Section 13.1(d).
- "Initial Applicable Term" has the meaning set forth in Section 2.1.
- "Intellectual Property" or "IP" means any of the following, whether subsisting now or in the future anywhere in the world: (a) patents and pending patent applications; (b) trademarks, service marks, trade names and trade dress, and associated goodwill and rights of publicity and

all rights associated therewith; (c) copyrights, including copyrights in works of authorship and computer Software; (d) confidential and proprietary information, including trade secrets and proprietary algorithms; (e) data base rights; (f) design rights and rights in designs; (g) rights in domain names; (h) rights in know-how; (i) all other intellectual property rights subsisting now or in the future, anywhere in the world; and (j) registrations, right to register and pending applications for registration of the foregoing, renewals, extensions, continuations, divisions or reissues thereof now or hereafter in force throughout the world (including rights in any of the foregoing); and (k) any and all causes of action arising from or related to any of the foregoing.

"Intellectual Property Rights" means any and all common law, statutory and other rights in Intellectual Property honored and/or enforceable under any Laws.

"Law" means any applicable order, writ, injunction, decree, judgment, ruling, statute, law, rule or regulation of any federal, state, municipal or local government, governmental authority, regulatory or administrative agency, governmental commission, department, board, bureau, court or tribunal or any arbitrator or arbitral body.

J"Lewisville, Texas Leased Real Property" means the approximately 78,413 square feet of real property designated as Building 3 of the project commonly referred to as Convergence Office Center in Lewisville, Texas.

"Migration Assistance Services" has the meaning set forth in Section 14.4(a).

["Minnesota Data Center" means the data center located in Eden Prairie, Minnesota that is referenced in the definition of the term "Minnesota Data Center Interest" in the APA.]

"New Developments" means new Systems implemented by Supplier that are substantially different from the existing Systems, including major replacements to existing Systems and any directly related major replacements to work processes, policies and procedures implemented by Supplier.

"Parties" means AFI and Ocwen [and Walter].

"**Person**" means any individual, partnership, firm, corporation, association, joint venture, limited liability company, trust or other entity, or any governmental entity.

"Post Term Services" has the meaning set forth in Section 2.3.

"Properties" has the meaning set forth in Section 15.17(a).

"Purchase Agreement" has the meaning set forth in Section 15.17(a).

- "Recipient Data" has the meaning set forth in Section 8.1.
- "Recipient Equipment" means all Equipment owned or leased by a Recipient and used in connection with the Services.
- "Recipient Facilities" has the meaning set forth in Section 6.1(a).
- "Recipient Indemnified Parties" has the meaning set forth in Section 13.1(b).
- "Recipient Software" means all Software owned by, or provided under license to, Recipient and used in connection with the Services (and all modifications, replacements, upgrades, enhancements, documentation, materials and media relating to the foregoing).
- "Recipient System" means an interconnected grouping of Recipient Equipment and/or Recipient Software used in connection with the Services, and all additions, modifications, substitutions, upgrades or enhancements thereto.
- "Relationship Manager" has the meaning set forth in Section 5.1.
- "Required Consents" means the consents, if any, required from third parties in connection with Supplier's provision, and Recipient's receipt, of the Services.
- "ResCap Legacy Subcontract" means an agreement between ResCap and a third party, pursuant to which ResCap uses such third party as a subcontractor for Reverse Services under the SSA.
- "ResCap Legacy Subcontractor" means a third party that entered into a ResCap Legacy Subcontract with ResCap.
- "Services" means all or any of AFI Services, Buyer Services, Additional Services, Customized Services, or Migration Assistance Services provided by a Supplier, as applicable given the context.
- "SSA" has the meaning set forth in the Recitals.
- "**Software**" means programs and programming (including the supporting documentation, media, on-line help facilities and tutorials).
- "Statement(s) of Work" means the mutually agreed internal documents that set forth the Services in respect of the Functional Service Areas that are identified in <u>Schedule A-1</u> or <u>Schedule A-2</u>, as applicable.
- "Subcontractors" means Supplier's contractors or other agents of Supplier that perform a portion of the Services.
- "Supplement" has the meaning set forth in Section 3.2.
- "Supplier" means (i) in the case of AFI Services, AFI and any AFI Affiliates that provide such Services, and (ii) in the case of Buyer Services, [Buyer], and any Affiliates of [Buyer] that

provide such Services. For purposes of clarity, all references to "Supplier" refer only to the Party that is acting as the "Supplier" with respect to the Equipment, Software and other applicable responsibilities for the Services being performed by such Party in its capacity as the "Supplier".

"Supplier Equipment" means Equipment owned or leased by the Party that is acting in its capacity as the "Supplier" hereunder, or an Affiliate or Subcontractor of such Party, and used in connection with the Services.

"Supplier Facilities" has the meaning given in Section 6.2.

"Supplier Indemnified Parties" has the meaning set forth in Section 13.1(a).

"Supplier Personnel" means those employees, representatives, contractors, Subcontractors and agents of the Party that is acting in its capacity as the "Supplier" hereunder, who perform any Services under this Agreement.

"Supplier Software" means all software programs and programming owned by, or provided under license to, the Party that is acting in its capacity as the "Supplier" and used to provide the Services (and all modifications, replacements, upgrades, enhancements, documentation, materials and media relating to the foregoing).

"Supplier System" means an interconnected grouping of Supplier Equipment and/or Supplier Software used in connection with the Services, and all additions, modifications, substitutions, upgrades or enhancements thereto.

"Systems" means Recipient Systems and Supplier Systems or any of them.

["Texas Data Center" means the data center located in Lewisville, Texas that is referenced in the definition of the term "Texas Data Center Interest" in the APA.]

"**Term**" means the period commencing on the Effective Date and continuing thereafter until the termination or expiration of the last Applicable Term.

"Third Party Claim" has the meaning set forth in Section 13.1(d)(1).

"Third Party Claim Notice" has the meaning set forth in Section 13.1(d)(1).

2. TERM AND RENEWAL.

2.1 <u>Initial Applicable Term.</u> Unless a Service is earlier terminated or extended in accordance with the terms of this Agreement, Supplier shall provide to Recipient each Service for the service period specified in <u>Schedule A</u>, or if no service period is so specified for a particular Service, for ninety (90) days after the Effective Date (such period in respect of a

particular Service, the "Initial Applicable Term").³

- 2.2 Extensions of Applicable Term. Each Party, in its capacity as a Recipient of a Service, will have the right to extend the Initial Applicable Term of such Service for a period of up-to- ninety (90) days, upon notice to Supplier at least thirty (30) days prior to the expiration of the Initial Applicable Term for such Service; provided, however, that (i) AFI shall have the right to extend for up-to- an additional twelve-twelve-(12six-(6)">twelve-twelve-(12six-(6)">twelve-twelve-(12six-(6)") month period the Initial Applicable Term for the applicable Service, and (ii) Buyer shall have the right to extend for up-to- an additional six (6) month period the Initial Applicable Term for any or all of the AFI Services in the ITG Functional Service Area, upon notice to AFI at least thirty (30) days prior to the expiration of the Initial Applicable Term for the applicable Service. No other Service extensions will be permitted except as otherwise mutually agreed upon in writing by the Parties. Each Initial Applicable Term is so extended is referred to herein as an "Extended Applicable Term"; collectively with the Initial Applicable Term for the particular Service, the "Applicable Term".
- 2.3 <u>Post-Term Service</u>. In the event that Recipient requests that Supplier continue to provide any Service past the end of its corresponding Applicable Term, Supplier may, in its sole discretion, continue to provide such Service (a "**Post Term Service**"), in which event this Agreement will be extended for purposes of such Post Term Services, <u>provided</u> that the Charges for such Post Term Services shall increase (over the Charges as of the expiration of the Applicable Term for such Post Term Service) by _____ (_%) per month plus any additional set-up, configuration, deployment or other reasonable costs and expenses to be incurred by Supplier as a result of the extension.

3. SERVICES.

3.1 Services.

(a) <u>Performance.</u> AFI will provide the Services within each of the Functional Service Areas listed in <u>Schedule A-1</u> (the "AFI Services"), and Buyer will provide the Services within each of the Functional Service Areas listed in <u>Schedule A-2</u> (the "Buyer Services"), in each case beginning on the Effective Date. Services provided by a Supplier under this Agreement may be provided by that Supplier directly or through any of its Affiliates, the Subcontractors listed on <u>Schedule F-1</u> or <u>F-2</u>, as the case may be, or any other Subcontractors requested by Supplier and, to the extent Supplier is required to obtain Recipient's approval pursuant to <u>Section 4.1(a)</u>, approved by Recipient, such approval not to be unreasonably withheld, conditioned or delayed. A Supplier will not be relieved of any of its obligations under this Agreement as a result of the provision of Services by any of Supplier's Affiliates or, subject to <u>Section 4.1(b)</u>, Supplier's Subcontractors or other Supplier Personnel pursuant to this <u>Section 3.1(a)</u>. AFI agrees that it will not enter into any new subcontracting arrangement that would

Note to Buyer: Schedule A will specify (i) a <u>twelve (12twelve (3) four (4)</u> month initial term for capital market services to be provided by Buyer to AFI, (ii) a <u>twelve (12twelve (12six (6))</u> month initial term for IT related services to be provided by Buyer to AFI, <u>((iii))</u> a six (6) month initial term for licensing services to be provided by Buyer to AFI, <u>((iii))</u> a six (6) month initial term for IT related services to be provided by AFI to Buyer.

involve the transfer to a Subcontractor or another third party of Buyer's personal information, without the prior written consent of Buyer. For purposes of the preceding sentence, a "new subcontracting arrangement" will not include any subcontract that is in place as of the Effective Date, or any renewals or extensions thereof. Buyer will not transfer to any Subcontractor or any other third party any of AFI's or its Affiliates' personal information (including customer information), without AFI's prior written consent, except that such consent shall not be required for transfers of personal information by Buyer to a Subcontractor performing services under a ResCap Legacy Subcontract listed on Schedule F-2. The Supplier Personnel providing Services will at all times be qualified to provide the Services assigned to them.

- (b) <u>Recipients and Facilities</u>. Supplier will provide the Services to Recipient at the Recipient Facilities for which such Services are provided as of the Effective Date or, with respect to any particular Services, such other locations as may be specifically identified in <u>Schedule D-1</u> or <u>Schedule D-2</u> with respect to such Services. Supplier will negotiate in good faith to provide Services in support of any additional Recipient Facility, but shall not be obligated to provide Services to additional Recipient Facilities unless the pricing and terms for such additional facilities has been mutually agreed upon by the Parties.
- Service Standards. Supplier will perform or cause to be performed the (c) Services referenced in the applicable Statement(s) of Work, to Recipient (i) with reasonable skill, care and diligence, and (ii) performed in substantially the same manner (including historic levels of service, historical usage levels and geographic provisioning) that such Services were generally performed by Supplier for Recipient immediately prior to the Effective Date (or, in the case of the Buyer Services, the same manner that such Services were generally performed by or on behalf of ResCap for AFI immediately prior to the Effective Date) and as Services are provided thereafter in substantially the same manner (including historic levels of service, historical usage levels and geographic provisioning) as Supplier generally performs such Services for its own businesses (and, in the case of the Buyer Services, at least in the same manner and levels of service that such Services were generally performed by or on behalf of ResCap for AFI immediately prior to the Effective Date and in the case of AFI Services, in a manner and at levels of service that are not materially less than those generally performed by or on behalf of AFI for ResCap under the SSA immediately prior to the Effective Date) in accordance with its thencurrent processes and procedures (except to the extent such Services differ because of the need to follow legal corporate formalities and to keep Recipient Data separate from Supplier data, or as may be mutually agreed through the Parties' change control principles pursuant to Sections 3.2 and 3.3). In no event will Supplier be required to make any customization to the Services (or Supplier's associated Systems or processes) that are unique to Recipient, except for customizations that are expressly agreed upon in accordance with Sections 3.2 and 3.3. Each Party acting in its capacity as Recipient shall, and such Party shall cause all of its Affiliates that are Recipients to, comply with any applicable terms and conditions of third party contracts used by Supplier in connection with the Services. Supplier will perform its obligations under this Agreement in accordance with applicable Law. However, except as may be expressly provided in a Statement of Work, Recipient shall be solely responsible for its own compliance with applicable Law, and nothing in this Agreement shall constitute or be construed as a representation or warranty by Supplier that any Service provided by Supplier is sufficient to satisfy any of Recipient's obligations under any applicable Law.

- (d) <u>Implied Services</u>. If any services, functions or responsibilities performed by Supplier (or, in the case of the Buyer Services, performed by or on behalf of ResCap) for Recipient as of the Effective Date are not specifically described in this Agreement but are necessary for Supplier to perform or provide the Services described in this Agreement in accordance with <u>Section 3.1(c)</u>, such services, functions or responsibilities shall be deemed to be included within the scope of the Services to the same extent and in the same manner as if specifically described in this Agreement, except to the extent otherwise specified in this Agreement (including any services, functions and responsibilities of Recipient under this Agreement but are necessary for Recipient to perform or provide such services, functions and responsibilities described in this Agreement). Except as otherwise expressly provided in this Agreement, as between the Parties, Supplier shall be responsible for providing the facilities, personnel and other resources as necessary to provide the Services consistent with the requirements of <u>Section 3.1(c)</u>.
- (e) <u>Not a Requirements Contract</u>. This Agreement will not be construed as a requirements contract and will not be interpreted to prevent Recipient from obtaining from third parties, or providing itself, any or all of the Services or similar services.
- (f) <u>Supplier Modifications</u>. Supplier may reasonably supplement, modify, substitute or otherwise alter any of the Services from time to time in a manner consistent with supplements, modifications, substitutions or alterations made for similar services provided or otherwise made available by Supplier or its Affiliates to itself or its Affiliates; <u>provided</u> that the Services continue to be provided in accordance with the standards set forth in <u>Section 3.1(c)</u> notwithstanding such supplements, modifications, substitutions or alterations and such supplements, modifications, substitutions or alterations shall not materially diminish the scope or quality of the Services.
- (g) No Expansion of Existing Rights. The Parties acknowledge and agree that this Agreement is intended to provide for the continuation of services that were previously provided under the SSA immediately prior to the Effective Date (to the extent such services are within the scope of the AFI Services and Buyer Services, as the case may be), but not expand the scope of such services or expand the group of Persons entitled to receive such services beyond AFI and the AFI Affiliates that are Recipients in the case of the Buyer Services or beyond the Acquired ResCap Businesses in the case of the AFI Services, except as otherwise agreed in writing by the Parties.
- 3.2 <u>Additional Services</u>. A Party may, from time to time during the Term, upon at least thirty (30) days prior written notice to the other Party, request that the other Party provide additional services, functions and responsibilities not within the scope of the Services provided by the performing Party ("Additional Services"). Upon such request, the Parties shall discuss in good faith the scope and nature of such request and related issues, including any potential additional Charges, it being understood and agreed that neither Party shall be obligated to agree to provide any Additional Services. Any such Additional Services agreed to by the Parties will be provided under supplements to <u>Schedule A-1</u> or <u>Schedule A-2</u>, as applicable, entered into by the Parties ("Supplements") for the Charges set forth therein and mutually agreed upon. Supplements shall be in the form of <u>Schedule E</u>. The Parties will use the process developed by

AFI and ResCap under the SSA to (a) submit and prioritize requests for Additional Services and (b) create and execute Supplements for Additional Services. Such processes include, without limitations, meetings between the transition leads appointed by each Party to review all change requests, processes for work with the appropriate business leads as well as any other necessary resources and processes for amending relevant SOWs as agreed upon by the Parties.

3.3 Changes to Services/Change Control Procedures.

- (a) <u>Customized Services</u>. Supplier shall not be required to customize Services or Supplier Systems for the applicable Recipient ("**Customized Services**") except as set forth in this **Section 3.3**.
- (b) <u>Changes Requested by Recipients</u>. Recipient from time to time, upon at least thirty (30) days notice, may require that Supplier decrease or reasonably increase the volume of the Services provided to Recipient and the Charges shall be modified in accordance with the procedures and minimum revenue commitments set forth in <u>Schedule C</u>. A Party may, from time to time during the Term, upon at least thirty (30) days prior written notice to the other Party, request that the other Party provide Customized Services. Upon such request, the Parties shall discuss in good faith the scope and nature of such request and related issues, it being understood and agreed that neither Party shall be obligated to agree to provide Customized Services. If the Parties mutually agree on the provision of Customized Services, such request for Customized Services and such Customized Services will be provided under Supplements to <u>Schedule A-1</u> or <u>Schedule A-2</u>, as applicable. Before Supplier is required to provide any Additional Services or Customized Services, the Parties shall jointly agree on the applicable Charges for any agreed Additional Services or Customized Services, including any Charges that may be required to equitably compensate Supplier for any additional costs it may reasonably incur in connection with any changes to the Services.
- (c) <u>Changes Required by Applicable Law.</u> If Customized Services are necessary in order to comply with applicable Laws or changes to Supplier's third party contracts, Supplier will provide such Customized Services unless (i) such Customized Services require a material change to a Supplier System, or the implementation of a new Supplier System, or (ii) providing such Customized Services is not practicable given the then-current characteristics of the Supplier Systems, and the use thereof for Supplier and its Affiliates, and subject to the Parties jointly agreeing on the applicable Charges for such Customized Services. Any such Customized Services will be provided under Supplements to <u>Schedule A-1</u> or <u>Schedule A-2</u>, as applicable. The Parties will negotiate in good faith in order to promptly agree on the applicable Charges for any such necessary Customized Services. If providing such Customized Services is not practicable given the Supplier Systems, Recipient may purchase such services from a third party, <u>provided</u> that the Parties must agree on (A) the activities required to transition the affected Services from Supplier to such third party, (B) the impact on the remaining Services, and (C) the Charges associated with removing such Services and (D) the Charges for the remaining Services.

⁴ Note to Buyer: The Parties to discuss appropriate minimum monthly revenue commitments for each Service provided under this Agreement (e.g., monthly service charges cannot be less than XX% of the initial monthly fees).

3.4 <u>Recipients' Obligations</u>. Supplier's failure to perform its obligations under this Agreement will be excused (and any rights of the Recipient arising as a consequence of such failure will not be exercised by the Recipient) if and to the extent such Supplier non-performance is caused by (a) the wrongful or tortious actions of Recipient or a third party contractor performing obligations on behalf of Recipient under this Agreement, or (b) the failure of Recipient or such a third party contractor to perform such Recipient's obligations under this Agreement. Recipient will be responsible for any additional costs incurred by Supplier in connection with providing the Services as a result of any such failure. Supplier will use commercially reasonable efforts to perform its obligations notwithstanding such failure, provided that Recipient works with Supplier through the Relationship Managers to remedy the failure.

3.5 Required Consents.

- (a) <u>Responsibility</u>. Each Party will be responsible for obtaining any Required Consents required under its own third party contracts pertaining to any Software, Equipment, Systems or other materials or associated services required in connection with the Services under this Agreement. Such responsibility shall include the administrative activities necessary to obtain the Required Consents. Each Party, in its capacity as "Recipient", will be responsible for paying for (or, if applicable, reimbursing the Party acting as "Supplier" for the payment of) the fees and expenses to the extent required to obtain such Required Consents to provide Services to Recipient, regardless of whether such Required Consents are required under third party contracts of Supplier or Recipient.
- (b) <u>Contingent Arrangements</u>. If, despite using commercially reasonable efforts, either Party is unable to obtain a Required Consent for which it is responsible under <u>Section 3.5(a)</u>, such Party will use commercially reasonable efforts to obtain a replacement license, product or right, as applicable. If such replacement is not or cannot be obtained using commercially reasonable efforts, the Parties will work together in good faith to develop a mutually acceptable alternative arrangement that is sufficient to enable Supplier to provide, and Recipient to receive, the Services without such Required Consent. The Party responsible for paying for the Required Consent (as provided in <u>Section 3.5(a)</u>) will be financially responsible for the costs of any such alternative arrangement. If such alternative arrangement cannot be agreed upon by the Parties or is required for a period longer than one hundred eighty (180) days following the Effective Date, either Party may require that the affected Services be discontinued in which case the Charges for Services will be equitably adjusted to account for such discontinued Services.
- 3.6 <u>Security Level; Additional Security Measures</u>. Supplier may take physical or information security measures (a) that affect the manner in which the Services are provided to maintain Supplier's current level (or, if greater, an industry-standard level) of physical and electronic security (including data security and data privacy) during the Term and (b) that address any new security-related issues, including compliance with applicable Law and governmental orders related to security and issues in connection with new technologies or threats. Supplier shall provide Recipient reasonable, prior written notice of any such physical or information security measures that are material to Supplier's delivery of the Services. Recipient shall provide all assistance reasonably requested by Supplier in connection with such security measures. Recipient shall pay to Supplier increased Charges for the applicable Services that may

be required to equitably compensate Supplier for any additional costs it may reasonably incur in connection with such security measures with respect to the Services provided to Recipient under this Agreement. For the avoidance of doubt, the Charges provided in $\underline{\text{Schedule C}}$ include the cost of Supplier providing the Services at the level of data security and data privacy required by AFI's information security policies in effect as of the Effective Date.

4. USE OF AFFILIATES AND SUBCONTRACTORS.

- 4.1 Affiliates and Subcontractors.
- (a) <u>Use of Affiliates and Subcontractors</u>. Subject to <u>Section 3.1(a)</u>, Supplier will have the right to use Affiliates and Subcontractors to assist Supplier in the provision of the Services. AFI, in its capacity as "Supplier", shall adhere to AFI's then-current "Third Party Vendor Management" policies (which shall not be substantially less comprehensive in scope than those policies in effect as of the effective date of the SSA) when engaging Subcontractors to perform AFI Services under this Agreement. Buyer, in its capacity as "Supplier", shall adhere to ResCap's "Third Party Vendor Management" policies in effect as of the effective date of the SSA, when engaging Subcontractors to perform Buyer Services under this Agreement. In respect of any material Services for which Supplier desires to engage a Subcontractor, Supplier shall seek and obtain Recipient's approval of such Subcontractor, which approval Recipient shall not unreasonably withhold, condition or delay; <u>provided</u>, <u>however</u>, that all Subcontractors listed on <u>Schedule F-1</u> or <u>F-2</u>, as the, case may be (and, in the case of the Buyer Services, all ResCap Legacy Subcontractors) are hereby deemed approved by Recipient.
- Supplier Responsibility for Affiliates and Subcontractors. Supplier shall (h) remain responsible for the Services performed by its Affiliates and Subcontractors to the same extent as if such Services were performed by such Supplier. However, notwithstanding anything to the contrary contained herein, in the event that any of Supplier's non-Affiliate Subcontractors fails to perform its obligations for which Supplier is utilizing it in connection with this Agreement, Supplier will not be responsible or liable for such failure. Supplier will exercise any rights that it may have under its contract with the applicable Subcontractor to cause the Subcontractor to resolve or cure any such failure within a reasonable time period in the same manner that Supplier responds to such failure with respect to the services such Subcontractor provides for Supplier or Supplier's Affiliates businesses, taking into consideration the circumstances of such failure, the nature of the services and the impact that such failure has on Recipient. If such failure is not resolved or cured within a reasonable time period, Supplier will exercise any rights that it may have under its contract with the Subcontractor in the same manner that Supplier responds to such failure with respect to the services such Subcontractor provides for Supplier or Supplier's Affiliates businesses, taking into consideration the circumstances of such failure, the nature of the services and the impact that such failure has on Recipient. Supplier shall be Recipient's sole point of contact regarding the Services, including with respect to payment. In addition, each Party, in its capacity as Supplier, will monitor and manage (including any necessary audits as to data privacy and security) its Subcontractors being used to perform Services for Recipient in compliance with Supplier's third party vendor management policies and procedures (or, in the case of the Buyer Services, ResCap's "Third Party Vendor Management" policies, in each case, in effect as of the effective date of the SSA), throughout the Term.

4.2 <u>Compliance</u>. Supplier will cause its employees and agents as well as the employees of its Affiliates and Subcontractors while at Recipient Facilities, to comply with the personnel, operational, safety and security procedures, policies, rules and regulations applicable to Recipient employees and agents and the Recipient Facilities, of which they have been given notice in advance by Recipient, or in the absence of such advance notice, with such procedures, policies, rules and regulations addressing similar issues in place at Supplier's facilities.

5. RELATIONSHIP MANAGEMENT.

- 5.1 <u>Relationship Managers</u>. Each Party will appoint an individual (each, a "**Relationship Manager**") who, from the Effective Date until replaced by the appointing Party, will serve as that Party's representative under this Agreement. Each Relationship Manager will (a) have overall responsibility for managing and coordinating the performance of the appointing Party's obligations under this Agreement, and (b) be authorized to act for and on behalf of the appointing Party concerning all matters relating to this Agreement. Neither Party will reassign a Relationship Manager, unless it provides reasonable prior written notice to the other Party. If a Relationship Manager ceases to be employed by the Party that appointed it or is reassigned by such Party, such Party will promptly appoint a new Relationship Manager and provide written notice to the other Party of the new Relationship Manager so appointed.
- 5.2 <u>Governance Model</u>. The Parties will conduct meetings and manage interactions in accordance with the governance model described in <u>Schedule B</u>.
- 5.3 Reports. Each Party will provide to the other Party the reports described in the applicable Statement of Work, in the format and at the frequencies specified therein.
- 5.4 <u>Other Reports and Information</u>. Each Party will provide the other Party access to such information and reports about ResCap or the Acquired ResCap Businesses (which information and reports are in the possession or under the control of such Party or any of its direct and indirect Affiliates), as such other Party may reasonably request from time to time for regulatory reporting, audit, risk management, compliance, corporate governance, bank and/or bank holding company supervision and/or examination or other business purposes.
- 5.5 <u>Regulatory Review.</u> Each Party will notify the other promptly of any formal request or order by a governmental agency or regulator or any internal or external audit examination or request to examine records regarding Recipient that are maintained by Supplier or to audit Supplier's performance of the Services, including, in each instance above, where Buyer acts as Supplier, the records maintained or possessed by Buyer regarding the assets purchased under the APA. Supplier will cooperate with any such examination or audit. Recipient will reimburse Supplier for the actual and reasonable out-of-pocket costs Supplier incurs in connection with any such examination or audit.
- 5.6 <u>Books and Records; Audit.</u> Each Party acting in its capacity as a Supplier will, and will cause its Affiliates providing Services hereunder to, keep books of account and other records, in reasonable detail and in accordance with generally accepted accounting principles, consistently applied, as to the Charges for providing the Services to Recipient pursuant to this Agreement, and shall make such books of account and other records available to Recipient (and

to any Governmental Authority that desires to audit Recipient) for inspection during normal business hours in a non-disruptive manner so as not to disrupt Supplier's business operations during the Term and for twenty-four (24) months thereafter for the purpose of performing audits and inspections of Supplier, related to the Services performed under this Agreement to verify the accuracy of Charges and invoices.

- 5.7 <u>Audit Assistance</u>. Each Party acting in its capacity as a Supplier shall, and will cause its Affiliates providing Services hereunder to, use commercially reasonable efforts to provide to such auditors, inspectors, regulators, and representatives, at Recipient's sole cost and expense, such assistance as they reasonably require. Recipient's auditors and other representatives shall comply with Supplier's security and confidentiality policies, procedures and requirements. Except for regulatory or governmental inspections or audits and except for audits by AFI of assets purchased under the APA or records related thereto, all such inspections or audits may be performed only by an independent third party auditing firm of national standing that has a written agreement with Recipient in which such third party agrees (i) to confidentiality obligations no less protective of Supplier than Recipient's confidentiality obligations under this Agreement and (ii) not to share with Recipient the Supplier information provided in connection with such inspection or audit, other than the findings and conclusions of the audit report.
 - 5.8 <u>Informal Dispute Resolution Procedures.</u> The informal dispute resolution procedures applicable to disputes under or in connection with this Agreement are set forth in **Schedule B**.
- 5.9 <u>Continued Performance</u>. Each Party agrees that it will, unless otherwise directed by the other Party, continue performing its obligations under this Agreement while any dispute is being resolved until this Agreement expires or is terminated in accordance with its terms, <u>provided, however</u>, that in the case of a dispute with regards to a Party's alleged failure to pay amounts in excess of two (2) times the average monthly Charges for the particular Services that are subject to such dispute and provided hereunder to such Party or its Affiliates, Supplier may suspend its performance of such Services until the earlier of such dispute is resolved or this Agreement is terminated; <u>provided</u>, <u>further</u>, that if Recipient pays such disputed amounts, (a) Supplier will continue to perform its obligations under this Agreement and (b) such payment will not constitute a waiver of any claims Recipient may have with respect to such disputed amounts.

6. FACILITIES.

6.1 <u>Use of Recipient Facilities</u>.

(a) General. Except as expressly set forth otherwise in any applicable Statement(s) of Work, each Party, acting in its capacity as "Recipient", will, and will cause its Affiliates to, provide to Supplier, at no charge, the space, office furnishings, janitorial service, telephone service, utilities (including air conditioning) and office-related equipment, supplies, and duplicating services at Recipient's premises that Supplier may reasonably need to provide the Services (collectively, the "Recipient Facilities"). Supplier's employees will have reasonable access to the Recipient Facilities twenty-four hours a day, seven days a week; provided, however, that in times of emergency or significant maintenance or construction activity, access may be restricted or denied if not required in connection with such emergency,

maintenance or construction. In such an event, Supplier will be excused from its performance of the Services to the extent Supplier is unable to provide the Services in accordance with the requirements of this Agreement as a result of such restricted access. [In addition to the foregoing, Buyer shall, at no additional Charge to AFI or its Affiliates, ensure that AFI, its Affiliates and Subcontractors will continue to have access and occupancy to all Recipient Facilities during each Applicable Term that are referenced in the SSA, in the same manner as such access was provided under the SSA, which access will be provided for at least an initial period of [twelve (12) months] beginning on the Effective Date, [and, at AFI's option, an additional twelve (12) month period commencing immediately after the expiration of such initial 12-month period. Notwithstanding the foregoing, except in respect of the Minnesota Data Center or the Texas Data Center, with respect to any facilities that Buyer assumes on the Effective Date, nothing herein will limit or restrict Buyer's right, on or after six (6) months after the Effective Date (but not earlier), to terminate, assign or sublease any facility lease or sell any such facility (in its entirety), upon at least 180 days prior written notice thereof to AFI, provided that the six (6) month limitation referenced above shall not limit or restrict Buyer's right to allow any facility lease for such a facility to expire within such six (6) month period so long as Buyer provides AFI at least 180 days prior written notice of such expiration (each of the foregoing rights to exclude, reject, terminate, assign, sublease, or allow to expire are collectively referred to herein as, a "Facility Termination Event"). AFI's access and occupancy rights to any facility to be provided by Buyer will be subject to a Facility Termination Event. However, for the avoidance of doubt, AFI will have no obligation to provide any AFI Services that are dependent upon AFI's access to facilities that are unavailable to AFI, whether pursuant to a Facility Termination Event or otherwise... [discuss]

- Relocation. If Recipient contemplates or makes a final decision to alter or relocate any of the Recipient Facilities that will require Supplier to relocate any of its personnel or Equipment from any Recipient Facility and the alteration or relocation could reasonably be expected to impact the Services (including the cost to perform, timing, ability to perform or level of performance), or Recipient desires Supplier to provide a Service to a new or different Recipient Facility or other location (each, a "Relocation/Alteration Request"), then Recipient will provide Supplier with sufficient advance notice of that fact to allow Supplier a reasonable amount of time to prepare for and implement the Relocation/Alteration Request as it impacts Supplier. The Parties will work together in good faith for a planned and orderly process for such Relocation/Alteration Request, timed as much as possible to coincide with the scheduled completion of migration of all of the applicable Services provided by each of them for the other from such facility, and each Party will bear its own relocation costs and expenses. Before requiring Supplier to complete a Relocation/Alteration Request, the Parties will agree on any adjustments to (i) the Services that may be required as a result of the Relocation/Alteration Request, and (ii) Supplier's Charges that may be required to equitably compensate Supplier for any additional costs it may reasonably incur in connection with the Relocation/Alteration Request.
- (c) <u>Supplier's Obligations</u>. Supplier will (i) keep the Recipient Facilities in good order, and (ii) not commit waste or damage to those facilities or use those facilities for any purpose other than providing the Services (and appropriate incidental use for internal Supplier administrative tasks unrelated to other customer accounts or Supplier marketing efforts).

(d) <u>Access to Recipient Systems</u>. Supplier will limit its, and will require all Supplier Personnel who have access to Recipient Systems to limit their, access to those portions of such Recipient Systems for which they are authorized in connection with the Services. Supplier will limit such access to those Supplier Personnel who are needed in order to provide the Services. Supplier will cooperate with Recipient in the investigation of any apparent unauthorized access by Supplier Personnel to the Recipient Systems.

6.2 Supplier Facilities and Systems.

- (a) <u>Supplier Facilities</u>. Supplier may perform the Services in such facilities maintained by or for Supplier or its Affiliates or Subcontractors (collectively, "**Supplier Facilities**") as Supplier reasonably deems appropriate, so long as appropriate security procedures have been implemented and are being observed at the Supplier Facilities. While at Supplier Facilities, each Party will, and will cause its and its Affiliates' personnel to, in its capacity as "Recipient", comply with Supplier's reasonable security requirements and other relevant policies of which they have been given notice, or in the absence of such notice, with such security requirements and other relevant policies addressing similar issues in place at Recipient's facilities.
- (b) Access to Supplier Systems. Each Party in its capacity as Recipient will limit its, and will require all Recipient personnel who have access to Supplier's computer or electronic data storage systems to limit their, access to those portions of such systems for which they are authorized in connection with their receipt and use of the Services. Each Party in its capacity as Recipient will (i) limit such access to those Recipient personnel who are authorized to use the Services, (ii) subject to Recipient's record retention policy, make available, upon Supplier's request, to Supplier a written list of the names of each individual who has been granted such access, and (iii) adhere to Supplier's security rules and procedures for use of Supplier's systems. All user identification numbers and passwords disclosed to the Recipients to permit Recipient personnel to access the Supplier systems will be deemed to be, and will be treated as, Supplier's Confidential Information. Recipient will cooperate with Supplier in the investigation of any apparent unauthorized access by Recipient personnel to Supplier's systems.
- 6.3 <u>Physical Security for Facilities</u>. Supplier will be responsible for all security procedures at any Supplier Facilities. Each Party as Recipient will provide all necessary security personnel and security equipment at the Recipient Facilities. While at the Recipient Facilities, each Party as Supplier will cause all Supplier Personnel to comply with Recipient's physical security procedures, as made known to Supplier's Relationship Manager, or in the absence of such notice, with such security procedures addressing similar issues in place at Supplier's facilities.

7. INTELLECTUAL PROPERTY AND PROPRIETARY RIGHTS.

7.1 Ownership of Intellectual Property. Except as expressly provided in Section 7.2, and 7.3, nothing in this Agreement shall grant or transfer any rights, title or interests in any Intellectual Property invented or created before or after the Effective Date by or on behalf of a Party and/or its Affiliates or otherwise controlled by or licensed to such Party and/or its Affiliates. A Party's use of the other Party's Intellectual Property shall not modify the ownership

rights set forth above.

- 7.2 <u>Development of Intellectual Property.</u> Subject to <u>Article 8</u>, as between the Parties (and without affecting Recipient's right, title and interest in and to Recipient Data or any Confidential Information of Recipient), all Intellectual Property developed or acquired by or for Supplier or any of its Affiliates in connection with providing the Services shall be owned by Supplier. Any services that rely on New Developments are outside the scope of the Services under this Agreement and therefore would have to be separately negotiated and agreed upon by the Parties.
- 7.3 <u>Limited License to Use Supplier Work Processes and Software</u>. Supplier grants to Recipient a limited, non-exclusive, non-assignable license, with the right for Recipient to grant sublicenses to its Affiliates, to use the work processes and Software owned by Supplier and/or its Affiliates that are provided to Recipient in connection with the Services solely to the extent necessary for Recipient to receive Services, and perform its responsibilities under this Agreement during the Term. THE SUPPLIER WORK PROCESSES AND SOFTWARE AND ALL DELIVERABLES ARE PROVIDED BY SUPPLIER ON AN AS-IS BASIS. SUPPLIER EXPRESSLY DISCLAIMS ANY REPRESENTATIONS OR WARRANTIES, EXPRESS OR IMPLIED, AS TO SUCH WORK PROCESSES AND SOFTWARE AND DELIVERABLES.
- 7.4 <u>No Implied Licenses</u>. Except as expressly specified in this Agreement, nothing in this Agreement will be deemed to grant to one Party, by implication, estoppel or otherwise, license rights, ownership rights or any other Intellectual Property Rights in any work processes, Software or other materials, data or information owned by the other Party or any Affiliate of the other Party.

8. RECIPIENT DATA.

- 8.1 <u>Definition</u>. The term "**Recipient Data**" means (i) any data or information of Recipient or its respective vendors, customers or other business partners that is provided to or obtained by Supplier solely in the performance of its obligations under this Agreement, including data and information regarding Recipient's businesses, customers, operations, facilities, products, consumer markets, assets and finances in whatever form or format (including in the form of any analysis or compilation of such data), and (ii) any data or information collected or processed solely in connection with the Services, even if such data or information is contained in reports, documentation, compilations or analyses provided to Recipient as part of the Services. For the avoidance of doubt, Supplier retains ownership of data pertaining to its performance of Services, including data pertaining to the volume and quality of the Services. For the avoidance of doubt, nothing in this Agreement shall be deemed to affect Supplier's or its Affiliates' use of data obtained outside of the performance of Supplier's obligations under this Agreement, even if that data includes data or information which is similar or identical to data or information included in the Recipient Data.
- 8.2 <u>Ownership</u>. As between Recipient and Supplier, Recipient owns and will continue to own all right, title and interest in and to all Recipient Data. To the extent that Recipient Data is embedded or incorporated into report templates owned by Supplier, Supplier will not be deemed to have assigned or transferred any of its right, title or interest in or to any

underlying Intellectual Property Rights thereto, but Supplier hereby grants to Recipient a perpetual, irrevocable, royalty free, non-transferable license to use, reproduce, display and perform (whether publicly or otherwise) and modify such report templates solely as necessary for Recipient's use in the ordinary course of its business. Recipient's use of such report templates is subject to the confidentiality obligations set forth below in **Article 9**. Additionally, Recipient's ownership of the Recipient Data reflected in such report templates shall not serve to transfer or otherwise affect any of Supplier's right, title and/or interest in and/or to any of underlying Intellectual Property Rights in any such report templates. Supplier may not use Recipient Data for any purpose except: (x) to provide the Services; (y) as required by AFI or Buyer to meet its financial and regulatory reporting requirements; or (z) as otherwise permitted under this Agreement, nor may Supplier sell, assign, lease or otherwise dispose of or commercially exploit Recipient Data. Supplier acknowledges that Recipient is not restricted hereunder from using or disseminating Recipient Data in the format in which it is provided by Supplier or any different format. For the avoidance of doubt, the restrictions on Supplier in this Section 8.2 shall not apply to any data or information generated by Supplier or its Affiliates that is of a technical or administrative nature to the extent relating generally to the operation of the Services infrastructure or to Supplier or its Affiliates' Intellectual Property, and not relating specifically and exclusively to the Recipients or based on or incorporating Recipient Confidential Information.

8.3 <u>Data Security</u>. Supplier will establish and maintain safeguards against the destruction, loss or alteration of Recipient Data in its possession that are no less rigorous than those for Supplier's operations. If Recipient reasonably requests additional safeguards for Recipient Data, Supplier will provide those additional safeguards as Additional Services under a new Supplement subject to <u>Sections 3.2</u> and <u>3.3</u>. Recipient has established backup security for data and keeps backup data files in its possession. If Recipient chooses to establish additional backup security and backup data files, it may do so, except that if such activities increase Supplier's cost of providing the Services or require Additional Services or Customized Services, such activities will be subject to <u>Sections 3.2</u> and <u>3.3</u>. In all instances, Recipient will ensure that Supplier will have access to the backup data files as Supplier reasonably needs to provide the Services.

9. CONFIDENTIALITY.

9.1 Obligations. Each Party will retain the other Party's non-public, proprietary and confidential information with the same degree of care as it uses to avoid unauthorized use, disclosure, publication or dissemination of its own confidential information of a similar nature ("Confidential Information") in confidence and not disclose the same to any third party nor use the same, except as expressly permitted in this Article 9. As used in this Agreement, "Confidential Information" shall include all non-public information, in any form, furnished or made available directly or indirectly by one Party to the other that is (a) marked confidential, restricted, proprietary and/or with a similar designation and/or (b) provided under circumstances reasonably indicating that it is confidential, restricted and/or proprietary. The terms and conditions of this Agreement, including the pricing for the Services and service descriptions and other details regarding service delivery, levels and performance, shall be deemed Confidential Information. In the case of either Party, Confidential Information also shall include, whether or not designated "Confidential Information", (i) Software, materials and other Intellectual Property

owned by the disclosing Party, and (ii) all non-public information concerning the operations, affairs and businesses of a Party or its Affiliates, the financial affairs of a Party or its Affiliates, and the relations of a Party or its Affiliates with its customers, employees and service providers (including customer lists, customer information, account information and consumer markets).

- 9.2 <u>Excluded Information</u>. Excepted from the obligations of confidence and non-use under this **Article 9** is that information, which:
- (a) is independently developed by the receiving Party without reference to Confidential Information of the disclosing Party;
 - (b) was, at the time of disclosure to it, in the public domain;
- (c) after disclosure to it, is published or otherwise becomes part of the public domain through no fault of the receiving Party;
- (d) was in the possession of the receiving Party at the time of disclosure to it without any obligation of confidentiality; or
- (e) was received after disclosure to it from a third party who had a lawful right to disclose such information to it without any obligation to restrict its further use or disclosure.
- 9.3 <u>Compelled Disclosure</u>. Notwithstanding the provisions of this <u>Article 9</u>, if the receiving Party becomes legally compelled to disclose any of the disclosing Party's Confidential Information, the receiving Party will, to the extent not prohibited by Law, promptly advise the disclosing Party of such legal requirement to disclose Confidential Information in order that the disclosing Party may seek a protective order at the disclosing party's own expense, may interpose an objection to such disclosure, take action to assure confidential handling of the Confidential Information, or take such other action as it deems appropriate to protect the Confidential Information in the circumstances. The receiving Party will disclose only that portion of the disclosing Party's Confidential Information that it is legally required to disclose.

9.4 Return of Information.

- (a) If expressly required under a Statement of Work or <u>Schedule A-3</u> or <u>A-4</u>, as applicable, Supplier will provide to Recipient a copy of the Recipient Data and Confidential Information of Recipient upon Recipient's written request or upon expiration or termination of this Agreement.
- (b) If not expressly required under a Statement of Work or <u>Schedule A-3</u> or $\underline{A-4}$, as applicable, Supplier will provide to Recipient a copy of the Recipient Data and Confidential Information of Recipient upon Recipient's written request, <u>provided</u> that (i) Supplier is able, using commercially reasonable efforts, to provide (and, if necessary with respect to commingled data, extract) such data, and (ii) Recipient compensates Supplier for the reasonable internal and out-of-pocket costs and expenses that are incurred by Supplier or its Affiliates in fulfilling such request and approved in advance by Recipient (such approval not to be unreasonably withheld or delayed).

- 9.5 Except for data and information for which a continuing license is granted to Supplier under this Agreement or under any other agreement to which Recipient or an Affiliate of Recipient is party, upon written request by Recipient, all of the Recipient Data and Confidential Information of Recipient in the possession of Supplier or its Affiliates will be destroyed by Supplier and certified as such to Recipient, without Supplier or its Affiliates retaining copies thereof, except that any instances of such Recipient Data or Confidential Information that are commingled with data or information that is owned or licensed by Supplier or its Affiliates ("Supplier Data") or in a form that is commercially impractical to destroy may be retained so long as Supplier does not access or make use of such Recipient Data or Confidential Information after receipt of the written request for destruction from Recipient; provided, however, that Supplier and its Affiliates shall be entitled to have access to and use of all Supplier Data that is commingled with such Recipient Data or Confidential Information and Supplier and its Affiliates shall be entitled to retain such Recipient Data and Confidential Information as required by Law, the recordkeeping requirements of any Governmental Authority or regulatory body and/or pursuant to Supplier's internal compliance policies, provided that Supplier and its Affiliates continue to handle all such retained Recipient Data and Confidential Information in accordance with the confidentiality and other provisions of this Agreement. Notwithstanding the foregoing, Supplier will not be required to destroy any Recipient Data or Confidential Information of Recipient that is required for Supplier to perform any unterminated Services for Recipient, and nothing contained in this Section 9.5 shall be construed to limit or affect any rights or licenses a Party would otherwise have in the absence of this Agreement.
- 9.6 Exception. Notwithstanding anything else in this Article 9, Supplier will have a right to disclose Recipient's Confidential Information to third parties to the extent reasonably necessary for Supplier to accomplish its responsibilities contemplated hereunder or for Supplier to comply with its obligations under applicable Law, including its reporting obligations under applicable Law, and, in the case of AFI, to use and disclose to AFI's Affiliates and to others Buyer Confidential Information relating to ResCap, to operate AFI's business, consistent with the manner such Confidential Information was used and disclosed by AFI prior to the Effective Date, and, in the case of Buyer, to use and disclose to Buyer's Affiliates and to others Confidential Information relating to ResCap, to operate an Acquired ResCap Business, consistent with the manner such Confidential Information was used and disclosed by ResCap prior to the Effective Date; provided, however, that such disclosure to third parties (other than AFI Affiliates or Buyer Affiliates, as applicable) will be made under confidentiality terms and conditions that are no less favorable to Recipient than the provisions of this Article 9 or, if less favorable, that are consistent with Recipient's customary practice for the nature of the third party receiving Recipient's Confidential Information.

9.7 Obligations.

- (a) Each Party's Confidential Information shall remain the property of that Party. Except as otherwise provided in this Agreement, each Party shall use at least the same degree of care, but in any event no less than a reasonable degree of care, to prevent disclosing to third parties the Confidential Information of the other as it employs to avoid unauthorized disclosure, publication or dissemination of its own information of a similar nature.
 - (b) In the event of any disclosure or loss of any Confidential Information of

the disclosing Party, the receiving Party shall notify the disclosing Party promptly upon becoming aware thereof.

- 9.8 <u>Loss of Confidential Information</u>. In the event of any disclosure or loss of any Confidential Information of the disclosing Party due to the fault of the receiving Party, the receiving Party shall promptly, at its own expense: (a) notify the disclosing Party in writing; and (b) cooperate in all reasonable respects with the disclosing Party to minimize the violation and any damage resulting therefrom.
- 9.9 <u>No Implied Rights.</u> Nothing contained in this <u>Article 9</u> shall be construed as obligating a Party to disclose its Confidential Information to the other Party, or as granting to or conferring on a Party, expressly or impliedly, any rights or license to the Confidential Information of the other Party.

10. COMPENSATION.

- 10.1 <u>General</u>. Buyer will pay to AFI the Charges as set forth in <u>Schedule C-1a</u> and elsewhere in <u>Schedule C</u>, and AFI will pay to [Buyer] the Charges as set forth in <u>Schedule C-2a</u> and elsewhere in <u>Schedule C</u>. Supplier will provide Recipient with invoices (with information linking the Services delivered to the invoice amounts) on a monthly basis on or before the last day of each calendar month for all Services performed by Supplier and all related Charges incurred by Recipient during that month. The last business day of the calendar month is the cut-off for delivered Services and related Charges to be invoiced in the next calendar month. With respect to any estimated Charges that are set forth on an invoice, Recipient will pay such estimated Charges as invoiced, and such estimated Charges will be adjusted in a subsequent invoice to reflect actual Charges as of the last day of the following month. Recipient will pay the invoices within thirty (30) days of the receipt of an invoice from Supplier. Any payment by Recipient is without prejudice of its right to contest the accuracy of any Charges.
- 10.2 <u>Taxes</u>. In addition to the prices determined pursuant to <u>Schedule C</u>, each Party will pay, and hold the other Party harmless against, all goods and services, sales, use, excise and other taxes, and other fees or assessments imposed by law in connection with the provision of the Services by the other Party, other than taxes measured by the other Party's net income. The Parties will cooperate with each other and use commercially reasonable efforts to assist the other in entering into such arrangements as the other may reasonably request in order to minimize, to the extent lawful and feasible, the payment or assessment of any taxes relating to the transactions contemplated by this Agreement; <u>provided</u>, <u>however</u>, that nothing in this <u>Section 10.2</u> will obligate Supplier to cooperate with, or assist, Recipient in any arrangement proposed by Recipient that would, in Supplier's sole discretion, have a detrimental effect on Supplier or any of Supplier's Affiliates.
- 10.3 <u>Invoicing and Payment</u>. Supplier will invoice Recipient in accordance with <u>Schedule C</u>. Payments for amounts past due will bear interest calculated on a per annum basis from the due date to the date of actual payment at a fluctuating interest rate equal at all times to the prime rate of interest announced publicly from time to time by Citibank, N.A. plus two percent (2%), but in no case higher than the maximum rate permitted by Law. Each Party will make payments under this Agreement by electronic funds transfer in accordance with payment

instructions provided by the other Party in its capacity as Supplier from time to time.

11. REPRESENTATIONS AND WARRANTIES.

- 11.1 <u>Services</u>. Supplier represents and warrants to Recipient that it will use the level of care in providing the Services required by <u>Section 3.1(c)</u>, or with respect to Services that are dedicated to Recipient (i.e., where Supplier does not provide similar services to itself or its Affiliates), Supplier will provide such Services using commercially reasonable efforts and in a workmanlike manner, and in accordance with commercially reasonable practices.
- 11.2 <u>Maintenance</u>. Supplier represents and warrants to Recipient that it shall provide for the maintenance of the Supplier Systems, Supplier Equipment and Supplier Software in the same manner that such functions were generally performed by Supplier for Recipient immediately prior to the Effective Date (or, in the case of Buyer, in the same manner that such functions were generally performed by or on behalf of ResCap for AFI under the SSA immediately prior to the Effective Date) and thereafter in a manner consistent with the standard of performance required by <u>Sections 3.1(c)</u> and <u>11.1</u>, for such items to generally operate in accordance with the manner in which they operated in the past or in the manner in which they operate in the future in support of Supplier and its Affiliates (and at least at the level that Systems, Equipment and Software operated under the SSA).
- 11.3 <u>Authorization</u>. Each Party represents and warrants to the other Party that: (a) it has the requisite corporate power and authority to enter into this Agreement and to carry out the transactions contemplated by this Agreement; and (b) the execution, delivery and performance of this Agreement and the consummation of the transactions contemplated by this Agreement have been duly authorized by the requisite corporate action on the part of such Party.
- 11.4 <u>Viruses</u>. Each Party represents and warrants to the other Party that it shall use the same efforts that it uses with respect to its own users and Systems, but in each instance, at least reasonable efforts, to avoid computer viruses from being introduced into the Systems of the other Party under this Agreement or the Systems that such Party is using to perform Services hereunder.
- 11.5 <u>Disclaimer</u>. EXCEPT AS EXPRESSLY SET FORTH IN THIS <u>ARTICLE 11</u>, NONE OF SUPPLIER, ITS AFFILIATES OR ANY OF THEIR RESPECTIVE REPRESENTATIVES MAKE OR HAVE MADE ANY REPRESENTATION OR WARRANTY, EXPRESS OR IMPLIED, AT LAW OR IN EQUITY, IN RESPECT OF THE SERVICES, INCLUDING WITH RESPECT TO (A) MERCHANTABILITY OR FITNESS FOR ANY PARTICULAR USE OR PURPOSE, (B) THE USE OF THE SERVICES BY RECIPIENT AFTER THE RECEIPT THEREOF, OR (C) THE PROBABLE SUCCESS OR PROFITABILITY OF RECIPIENT'S OR ANY OF RECIPIENT'S BUSINESS AFTER THE RECEIPT OF THE SERVICES.

12. INSURANCE.

12.1 <u>Coverages.</u> At all times during the Term and any extension hereof, both Parties will procure and maintain, at their own expense and for their own benefit, Comprehensive/Commercial General Liability insurance (including Professional Liability and

Contractual Liability coverages) with a bodily injury, death and property damage combined single limit of not less than \$2,000,000 per occurrence and in the aggregate. Each policy of insurance to be maintained hereunder will name the other Party, including its Affiliates, and the officers, directors and employees of each, as additional insureds.

In addition, both Parties will maintain in full force and effect during the Term the following insurance coverage.

- (i) Comprehensive Automobile Liability insurance covering owned, hired and non-owned vehicles with minimum limits of \$1,000,000 per person and \$1,000,000 per occurrence for bodily injury and \$1,000,000 property damage or combined single limit of \$1,000,000.
- (ii) Workers' Compensation insurance with limits as required by the laws of the states in which the Party's employees are employed, and Employer's Liability insurance with minimum limit of \$1,000,000 per occurrence.
- 12.2 <u>Policies</u>. Upon request, no more often than on an annual basis, both Parties will furnish the other with a certificate(s) from the insurance carrier(s) (having a minimum AM Best rating of A-) showing the coverages set forth above to be in force and effect. The insurance to be maintained hereunder will contain no more than a customary and reasonable deductible. Such insurance may be cancelled or materially altered only after thirty (30) days written notice is given to the other Party.
- 12.3 Risk of Loss. Supplier will be responsible for the risk of loss of, or damage to, any property of Recipient at a Supplier facility, unless and to the extent that such loss or damage was caused by the acts or omissions of Recipient and/or Recipient's Affiliates, and/or its and/or their employees, agents and/or Subcontractors. Recipient will be responsible for the risk of loss of, or damage to, any property of Supplier at a Recipient facility unless and to the extent that such loss or damage was caused by the acts or omissions of Supplier and/or Supplier's Affiliates, and/or its and/or their employees, agents and/or Subcontractors. The risk of loss of, or damage to, property in transit will remain with the Party arranging the shipment.

13. INDEMNIFICATION AND LIMITATIONS ON LIABILITY.

13.1 Indemnification.

(a) <u>Indemnification by Recipient</u>. Recipient will indemnify, hold harmless and defend Supplier and its Affiliates and their respective directors, officers, members, managers and employees (collectively, the "Supplier Indemnified Parties"), from and against any and all liabilities, damages, penalties, judgments, assessments, losses, costs and expenses in any case, whether arising under strict liability or otherwise (including reasonable outside attorneys' fees) (collectively, "Damages") suffered or otherwise incurred due to a Third Party Claim arising from or out of or relating to this Agreement, including the performance (or failure to perform) by Supplier of its obligations under this Agreement; <u>provided</u>, <u>however</u>, that to the extent and in the proportion Damages also arise out of or relate to the gross negligence or willful misconduct of any Supplier Indemnified Party, then the indemnity under this <u>Section 13.1(a)</u> will not apply.

(b) <u>Indemnification by Supplier</u>. Supplier will indemnify, hold harmless and defend Recipient and its Affiliates and their respective directors, officers, members, managers and employees (collectively, the "**Recipient Indemnified Parties**") from and against any and all Damages suffered or otherwise incurred due to a Third Party Claim to the extent arising from or out of or relating to the gross negligence or willful misconduct of any Supplier Indemnified Parties; <u>provided</u>, <u>however</u>, that to the extent and in the proportion Damages also arise from or out of or relate to the performance (or failure to perform) by any Recipient Indemnified Party of any obligations under this Agreement, then Supplier's indemnity under this <u>Section 13.1(b)</u> will not apply.

(c) <u>Infringement</u>.

- (1) <u>Indemnity</u>. Supplier will indemnify, hold harmless and defend the Recipient Indemnified Parties, and Recipient will indemnify, hold harmless and defend the Supplier Indemnified Parties, from and against any and all Damages due to a Third Party Claim to the extent that the claim alleges that any information, services or materials provided by the Indemnifying Party to the Indemnified Party under this Agreement infringes or misappropriates any Intellectual Property Right of such third party in any country in which Services are performed or received under this Agreement.
- (2) <u>Exclusions</u>. Neither Party shall have any obligation or liability to the other Party to the extent any infringement or misappropriation is caused by:
 - (i) modifications to any such information or materials made by the other Party, its Affiliates, or its third party contractors;
 - (ii) any combination of the allegedly infringing information or materials with items not provided by the Indemnifying Party, unless such combination was approved or directed in writing by the Indemnifying Party;
 - (iii) a breach of this Agreement by the other Party;
 - (iv) the failure of the other Party to use corrections or modifications provided by the Indemnifying Party offering equivalent features and functionality to the extent the Indemnifying Party notified the other Party that the failure to do so could result in infringement liability;
 - (v) the content provided by the other Party and not resulting from the Indemnifying Party's modification of that content without the other Party's approval; or
 - (vi) third party Software, except to the extent that such infringement or misappropriation arises from the failure of the Indemnifying Party to obtain the necessary licenses or to abide by the limitations of the applicable third party Software licenses.

- **(3)** Third Party Software Indemnities. As specified in Section 13.1(c)(2)(vi), the foregoing infringement indemnification does not apply to third party Software. With respect to third party Software provided by Supplier or its Subcontractors pursuant to this Agreement, Supplier and its Subcontractors shall use commercially reasonable efforts to obtain intellectual property indemnification for Recipient (or obtain intellectual property indemnification for itself and enforce such indemnification on behalf of Recipient) from the suppliers of such Software comparable to the intellectual property indemnification generally available in the industry for the same Software products. With respect to third party Software provided by Recipient pursuant to this Agreement, Recipient shall use commercially reasonable efforts to obtain intellectual property indemnification for Supplier and its Affiliates (or obtain intellectual property indemnification for itself and enforce such indemnification on behalf of the Supplier and its Affiliates) from the suppliers of such Software comparable to the intellectual property indemnification generally available in the industry for the same Software products.
- (d) <u>Indemnification Procedure</u>. The Party or other Persons making a claim for indemnification under this <u>Section 13.1</u> is referred to as the "<u>Indemnified Party</u>" and the Party or other Persons against whom such claims are asserted under this <u>Article 13</u> are referred to as the "<u>Indemnifying Party</u>". All claims by any Indemnified Party under this <u>Article 13</u> will be asserted and resolved as follows:
 - (1) In the event that any right, demand, claim, action and cause of action, assertion, notice of claim or assertion, complaint, litigation, suit, proceeding, formal investigation, inquiry, audit or review of any nature, civil, criminal, regulatory, administrative or otherwise, or any grievance or arbitration (each, a "Claim") is asserted or instituted in writing by any person or entity other than the Parties or their Affiliates that could give rise to Damages for which an Indemnifying Party could be liable to an Indemnified Party under this Agreement (such Claim, a "Third Party Claim"), the Indemnified Party will promptly send to the Indemnifying Party a written notice specifying the nature of such Third Party Claim, together with all information reasonably available to the Indemnified Party with respect to such Third Party Claim (a "Third Party Claim Notice"); provided, however, that a delay in notifying the Indemnifying Party will not relieve the Indemnifying Party of its obligations under this Agreement, except to the extent that such failure will have caused actual prejudice to the Indemnifying Party.
 - (2) In the event of a Third Party Claim, the Indemnifying Party will have ten (10) business days after receipt of the Third Party Claim Notice relating to such Third Party Claim to elect to undertake, conduct and control, through counsel of its own choosing (provided that such counsel is reasonably acceptable to the Indemnified Party) and at its own expense, the settlement or defense of such Third Party Claim (in which case the Indemnifying Party will not thereafter be responsible for the fees and expenses of any separate counsel retained by any Indemnified Party except as set forth below). Notwithstanding an Indemnifying

Party's election to appoint counsel to represent an Indemnified Party in connection with a Third Party Claim, an Indemnified Party will have the right to employ separate counsel, and the Indemnifying Party will bear the reasonable fees, costs and expenses of such separate counsel if (i) the use of counsel chosen by the Indemnifying Party to represent the Indemnified Party would present such counsel with a conflict of interest, or (ii) the Indemnifying Party will not have employed counsel to represent the Indemnified Party within a reasonable time after notice of the institution of such Third Party Claim. If the Indemnifying Party elects to undertake such defense, it will promptly assume and hold such Indemnified Party harmless from and against the full amount of any Damages resulting from such Third Party Claim to the extent provided herein. If the Indemnifying Party elects to undertake such defense, (x) the Indemnified Party agrees to cooperate with the Indemnifying Party and its counsel in contesting such Third Party Claim, and, if appropriate and related to such Third Party Claim, the Indemnifying Party and the Indemnified Party will reasonably cooperate with each other in connection with defending such Third Party Claim, and (y) such Third Party Claim may not be settled or compromised by the Indemnifying Party without the prior written consent of the Indemnified Party, which consent shall not be unreasonably withheld or delayed; provided, however, that in the event any Indemnified Party settles or compromises or consents to the entry of any judgment with respect to any Third Party Claim without the prior written consent of the Indemnifying Party, such Indemnified Party will be deemed to have waived all rights against the Indemnifying Party for indemnification under this Article 13. If the Indemnifying Party does not undertake the defense of such Third Party Claim, the Indemnified Party will have the right to contest, settle, compromise or consent to the entry of any judgment with respect to such Third Party Claim, and, in doing so, will not thereby waive any right to recourse therefor pursuant to this Agreement; provided, however, that at any time during the course of the matter the Indemnifying Party may assume the defense of such Third Party Claim by written notice of the same to the Indemnified Party.

(3) In the event of a Third Party Claim, from and after the delivery of a Claim Notice under this Agreement, at the request of the Indemnifying Party, the Indemnified Party will grant the Indemnifying Party and its representatives all reasonable access to the books, records and properties of such Indemnified Party to the extent reasonably related to the matters to which the Claim Notice relates. All such access will be granted during normal business hours and will be granted under conditions that will not unreasonably interfere with the businesses and operations of such Indemnified Party.

13.2 <u>Limitations on Liability</u>.

(a) Subject to the specific provisions and limitations of this Section 13.2, it is the intent of the Parties that each Party will be liable to the other Party for all Damages incurred by the non-breaching Party arising out of the breach of or failure to perform any of the breaching Party's representations, warranties, agreements, covenants or obligations under this Agreement. If a Party has a Claim for such Damages, it will promptly send to the breaching Party a written

notice specifying the nature of such Claim, together with all information reasonably available to such non-breaching Party with respect to such Claim (a "Claim Notice"); provided, however, that a delay in notifying the breaching Party will not relieve the breaching Party of its obligations under this Agreement, except to the extent that such failure will have caused actual prejudice to the breaching Party. In the event of such a Claim, the breaching Party will notify the non-breaching Party within forty-five (45) days of receipt of a Claim Notice whether or not the breaching Party disputes such Claim.

- (b) The total aggregate liability of Supplier under or in connection with this Agreement, regardless of the form of the action or the theory of the recovery, will be limited to:
 - With respect to Damages arising out of or relating to the failure by Supplier to meet the standards set forth in Section 11.1, subject to Section 13.2(b)(2), at Recipient's request, which request Recipient shall exercise in its good faith and reasonable discretion after reasonably consulting with Supplier and after exhausting the escalation process set forth in **Schedule B**, the replacement of (if applicable) or re-performance of, or repayment of the price paid for, the applicable Service during the period such Service did not meet the standards set forth in **Section 11.1**; provided, however, that if Recipient requests the replacement of or reperformance of such Service, Supplier shall, instead of such replacement or re-performance, have the option of refunding to Recipient the portion of the Charges actually paid to and retained by Supplier for such Service during the period such Service did not meet the standards set forth in Section 11.1 in those instances where re-performance or replacement of the Services cannot, in Supplier's good faith opinion, be provided or completed in a commercially reasonable manner. In the limited instances where Supplier determines that Supplier cannot replace or re-perform the Services in a commercially reasonable manner, (A) Recipient shall have the option to terminate the Applicable Term for such Service and not continue to pay for those specific Services without cost to Supplier other than the refund for the Services not performed, in accordance with the immediately preceding sentence, and (B) Supplier's obligation will be limited to providing the refund specified in the immediately preceding sentence and any reasonable cooperation requested by Recipient and at Recipient's expense in the transition of the terminated services to Recipient or a third party.
 - (2) With respect to indemnity Claims under Section 13.1 or other Claims arising out of or relating to the gross negligence or willful misconduct of Supplier, the fees actually received by Supplier from Recipient for Services hereunder during the twelve (12) months preceding the last act or omission giving rise to such Claims or, in the event such last act or omission occurs during the first twelve (12) months following the date hereof, an amount equal to twelve (12) times the fees paid for Services in the month preceding such last act or omission; and

- (3) With respect to all other Damages under or in connection with this Agreement, the fees actually received by Supplier from Recipient for Services hereunder during the twelve (12) months preceding the last act or omission giving rise to such Damages or, in the event such last act or omission occurs during the first twelve (12) months following the date hereof, an amount equal to twelve (12) times the fees paid for Services in the month preceding such last act or omission.
- (c) Each Party will use its commercially reasonable efforts to mitigate Damages for which it seeks recourse hereunder, including by promptly pursuing recovery under available insurance policies; <u>provided</u>, <u>however</u>, that the failure of such Party to successfully mitigate such Damages will not affect such Party's right to seek recourse with respect to such Damages so long as such Party will have used its commercially reasonable efforts to mitigate.
- (d) Any amounts payable under <u>Article 13</u> will be calculated after giving effect to any proceeds received from insurance policies covering the Damages.
- IN NO EVENT SHALL EITHER PARTY, ITS AFFILIATES OR ANY OF ITS OR THEIR DIRECTORS, OFFICERS, MEMBERS, MANAGERS, EMPLOYEES OR AGENTS BE RESPONSIBLE OR LIABLE TO THE OTHER PARTY OR ITS AFFILIATES FOR ANY INDIRECT, INCIDENTAL, CONSEQUENTIAL, SPECIAL, EXEMPLARY, OR PUNITIVE DAMAGES, OR FOR ANY LOSS OF PROFITS, LOSS OF REVENUE, LOSS RESULTING FROM INTERRUPTION OF BUSINESS OR LOSS OF USE OR DATA, EVEN IF THAT PARTY, ITS AFFILIATES OR ANY OF THEIR DIRECTORS, OFFICERS, MEMBERS, MANAGERS, EMPLOYEES OR AGENTS HAS BEEN ADVISED OF THE POSSIBILITY OF SUCH DAMAGES AND NOTWITHSTANDING ANY FAILURE OF ESSENTIAL PURPOSE OF ANY LIMITED REMEDY OF ANY KIND, UNDER ANY CONTRACT, NEGLIGENCE, STRICT LIABILITY OR OTHER THEORY, ARISING OUT OF OR RELATING IN ANY WAY TO THIS AGREEMENT OR ITS IMPLEMENTATION; PROVIDED, THAT THE FOREGOING SHALL NOT APPLY WITH RESPECT TO THIRD PARTY CLAIMS FOR DAMAGES FOR WHICH A PARTY IS INDEMNIFIED PURSUANT TO SECTION 13.1 TO THE EXTENT SUCH DAMAGES ARE AWARDED TO THE THIRD PARTY BRINGING SUCH CLAIM BY A COURT OF COMPETENT JURISDICTION.
- 13.3 Indemnification and Limitations on Liability Relating to Negligence and Strict Liability. ALL INDEMNITIES AND LIMITATIONS ON LIABILITY CONTAINED IN THIS ARTICLE 13 WILL APPLY WHETHER OR NOT THE INDEMNITEE OR PARTY CLAIMING DAMAGES WAS OR IS CLAIMED TO BE PASSIVELY, CONCURRENTLY OR ACTIVELY NEGLIGENT, AND REGARDLESS OF WHETHER LIABILITY WITHOUT FAULT IS IMPOSED OR SOUGHT TO BE IMPOSED ON SUCH INDEMNITEE OR PARTY.
- 13.4 <u>Exclusive Remedy</u>. Neither Party will have any remedy arising out of or relating to the breach of this Agreement other than the indemnification remedies set forth in <u>Section 13.1</u>

and Claims for direct damages referred to in <u>Section 13.2</u>, in each case subject to the limitations and exclusions set forth in this <u>Article 13</u>.

13.5 <u>Insurance</u>. Each Party will cause its insurers to waive their rights of subrogation against the other Party with respect to any Claim.

14. TERMINATION.

- **14.1** <u>Termination</u>. Without limiting the rights of the Parties under any other provision of this Agreement, this Agreement or any Service to be provided under this Agreement may be terminated as follows:
 - (a) by either Party acting in its capacity as "Supplier", upon written notice to the other Party if, after Recipient fails to pay the Charges for such Service when due in accordance with this Agreement, the terminating Party sends to the other Party an initial notice of such failure and the other Party fails to pay such Charges within thirty (30) days of receipt of such initial notice, <u>provided</u> that such termination will be only with respect to the particular Service(s) for which such Charges have not been paid;
 - (b) by either Party acting in its capacity as "Supplier", upon written notice to the other Party if, following a material breach by Recipient of this Agreement other than a payment default covered by clause (a) above, the terminating Party sends to the other Party a notice of such material breach and the other Party fails to cure such material breach within thirty (30) days; provided, however, that if cure cannot reasonably be accomplished within such 30-day period, neither this Agreement nor any Service may be terminated by reason of such breach for so long as the other Party commences a cure within such 30-day period and pursues such cure diligently to completion and such completion occurs within ninety (90) days of such written notice;
 - (c) by either Party acting in its capacity as "Recipient", upon written notice to the other Party if, following a material breach by Supplier of this Agreement, the terminating Party sends to the other Party an initial notice of such material breach and the other Party fails to cure such material breach within 30 days of receipt of such initial notice; provided, however, that if cure cannot reasonably be accomplished within such 30-day period, neither this Agreement nor any Service may be terminated by reason of such breach for so long as the other Party commences a cure within such 30-day period and pursues such cure diligently to completion and such completion occurs within ninety (90) days of such written notice, provided that such termination will be only with respect to the Service(s) provided to such Party acting in its capacity as "Recipient" that are identified in the written notice;
 - (d) by either Party acting in its capacity as "Recipient", upon at least thirty (30) days prior written notice to the other Party, <u>provided</u> that such termination will be only with respect to the particular Service(s) provided to such Party acting in its capacity as "Recipient" that are identified in the written notice: or
 - (e) otherwise upon mutual agreement of the Parties.

- 14.2 <u>Election of Terminating Party</u>. In connection with any termination pursuant to this <u>Article 14</u>, the Party exercising such termination rights may elect to continue to receive the Services (other than Services so terminated) that the other Party provides to it.
- 14.3 <u>Survival</u>. Any provision of this Agreement which contemplates performance or observance subsequent to any termination or expiration of this Agreement will survive any termination or expiration of this Agreement and continue in full force and effect including, but not limited to, the following: <u>Articles 7</u> (Intellectual Property and Proprietary Rights) (excluding <u>Section 7.3</u>), <u>8</u> (Recipient Data), <u>9</u> (Confidentiality), <u>10</u> (Compensation), <u>11</u> (Representations and Warranties), <u>12</u> (Insurance), <u>13</u> (Indemnification and Limitations on Liability), and <u>Sections 5.4</u> (Other Reports and Information), <u>5.5</u> (Regulatory Review), <u>5.6</u> (Books and Records; Audit), <u>5.7</u> (Audit Assistance), <u>14.3</u> (Survival), <u>15.3</u> (Counterparts), <u>15.5</u> (Force Majeure), <u>15.6</u> (Further Assurances), <u>15.7</u> (Governing Law; Jurisdiction and Forum; Waiver of Jury Trial), <u>15.8</u> (Headings), <u>15.11</u> (Public Announcements), <u>15.13</u> (Severability), <u>15.14</u> (No Third Party Beneficiaries) <u>and</u>), <u>15.16</u> (Non-Solicitation) and <u>15.17</u> (Additional Agreements Relating to Certain Properties).

14.4 <u>Migration Assistance Services</u>.

- (a) Supplier shall use commercially reasonable efforts to provide, or shall cause a third party service provider to use commercially reasonable efforts to provide, to Recipient the migration services identified as such in <u>Schedule A</u> (the "Migration Assistance Services") in connection with the transition to Recipient of Supplier's performance of the applicable Services (each, a "Transition"). In the absence of such Migration Assistance Services on <u>Schedule A-3</u> or <u>A-4</u>, as applicable, the Parties shall negotiate in good faith to agree upon the Migration Assistance Services to be provided as Additional Services pursuant to <u>Section 3.2</u>.
- (b) The Migration Assistance Services may include the following, among other Services: (i) developing, together with Recipient, a plan for the orderly Transition, including the migration of Recipient Data from Supplier to Recipient or to a third party designated by Recipient; (ii) providing reasonable training for personnel of Recipient in the performance of the Services then being transitioned to Recipient or the third party designated by Recipient, to the extent Services are being assumed by that third party; and (iii) providing cooperation to Recipient to facilitate the Transition, in accordance with a mutually agreed upon transition plan for the applicable Services. If Supplier provides any access to Supplier facilities, Systems or resources in connection with the execution of such plan or any other Migration Assistance Services, Recipient will, and will cause its personnel and any third parties having such access to, comply with Supplier's security and confidentiality requirements. Notwithstanding the foregoing, Supplier shall not be required to grant access to or share any of Supplier's proprietary information, data or technology with any third party that is not a Recipient.

15. GENERAL.

15.1 Acknowledgement. The Parties acknowledge that the terms and conditions of this Agreement have been the subject of active and complete negotiations, and that this Agreement will be construed as if drafted jointly by the Parties and no presumption or burden of proof will arise favoring or disfavoring either Party by virtue of the authorship of any provision of this

Agreement.

- 15.2 <u>Binding Effect; No Assignment</u>. This Agreement is binding upon and inures to the benefit of the Parties and their respective successors, permitted assigns and legal representatives. This Agreement is not assignable by either Party without the prior written consent of the other Party, and any other purported assignment will be null and void.
- **15.3** Counterparts. This Agreement may be executed in two or more counterparts, each of which will be deemed an original, but all of which will be considered one and the same agreement, and will become effective when one or more counterparts have been signed by each of the Parties and delivered (including by facsimile) to the other Parties.
- 15.4 Entire Agreement. This Agreement, including the Schedules hereto, which are each hereby incorporated herein and made a part hereof, contains the entire agreement between the Parties with respect to the subject matter hereof. While purchase orders, invoices or similar routine documents may be used to implement or administer provisions of this Agreement, any provisions of such documents that add to, vary, modify or are at conflict with the provisions of this Agreement will be deemed deleted and will have no force or effect on either Party's rights or obligations under this Agreement.

15.5 Force Majeure.

- (a) "Force Majeure Event" means any event beyond the reasonable control of the Party affected that significantly interferes with the performance by such Party of its obligations under this Agreement, including acts of God, strikes, lockouts or industrial disputes or disturbances, civil disturbances, arrests or restraint from rulers or people, interruptions by government or court orders or present and future valid orders of any regulatory body having proper jurisdiction (other than any such interruption arising from the failure by the Party claiming force majeure to comply with any applicable regulation or to obtain and comply with any required permit), acts of the public enemy, wars, riots, blockades, insurrections, inability to secure labor or secure materials upon terms deemed practicable by the Party affected (including by reason of allocations, voluntary or involuntary, promulgated by authorized governmental agencies), epidemics, landslides, lightning, earthquakes, fire, storm, floods, washouts, explosions, breakage or accident to machinery.
- (b) If a Force Majeure Event is claimed by either Party, the Party making such claim will orally notify the other Party as soon as reasonably possible after the occurrence of such Force Majeure Event and, in addition, will provide the other Party with written notice of such Force Majeure Event within two (2) business days after the occurrence of such Force Majeure Event.
- (c) Except for a Party's obligations to make payments hereunder (including those obligations of Section 10.3), neither Party will be liable for any nonperformance or delay in performance of the terms of this Agreement when such failure is due to a Force Majeure Event and the Charges payable by Recipient shall be equitably adjusted such that Recipient is not obligated to pay any Charges for any Services to the extent not performed due to a Force Majeure Event. If either Party relies on the occurrence of a Force Majeure Event as a basis for

being excused from performance of its obligations hereunder, such Party relying on the Force Majeure Event will (i) provide an estimate of the expected duration of the Force Majeure Event and its probable impact on performance of such Party's obligations hereunder and (ii) provide prompt notice to the other Party of the cessation of the Force Majeure Event.

- (d) Upon the occurrence of a Force Majeure Event, the same will, so far as possible, be remedied as expeditiously as possible using commercially reasonable efforts. It is understood and agreed that nothing in this **Section 15.5(d)** will require the settlement of strikes, lockouts or industrial disputes or disturbances by acceding to the demands of any opposing party therein when such course is inadvisable in the discretion of the Party having the difficulty.
- **15.6** <u>Further Assurances</u>. Each Party covenants and agrees that, subsequent to the execution and delivery of this Agreement and without any additional consideration, each Party will execute and deliver any further legal instruments and perform any acts that are or may become necessary to effectuate the purposes of this Agreement.

15.7 Governing Law; Jurisdiction and Forum; Waiver of Jury Trial.

- (a) This Agreement will be governed by and construed in accordance with the laws of the State of New York without reference to the choice of law principles thereof.
- (b) Each Party irrevocably submits to the jurisdiction of any New York state court or any federal court in the State of New York, in each case sitting in the Southern District of Manhattan, in any action arising out of or relating to this Agreement, and hereby irrevocably agrees that all claims in respect of such action may be heard and determined in such New York state or federal court. Each Party hereby irrevocably waives, to the fullest extent it may effectively do so, the defense of an inconvenient forum to the maintenance of such action or proceeding. The Parties further agree, to the extent permitted by law, that final and unappealable judgment against any of them in any proceeding contemplated above will be conclusive and may be enforced in any other jurisdiction within or outside the United States by suit on the judgment, a certified copy of which will be conclusive evidence of the fact and amount of such judgment.
- (c) Each Party waives, to the fullest extent permitted by Law, any right it may have to a trial by jury in respect of any action, suit or proceeding arising out of or relating to this Agreement. Each Party certifies that it has been induced to enter into this Agreement by, among other things, the mutual waivers and certifications set forth above in this **Section 15.7**.
- 15.8 <u>Headings</u>. The Article and Section headings contained in this Agreement are inserted for convenience of reference only and do not affect the meaning or interpretation of this Agreement. All references to Article or Sections contained herein mean Article and Sections of this Agreement, as the case may be, unless otherwise stated and except in the Schedules hereto, wherein references to Articles or Sections mean Articles and Sections of such Schedule, as the case may be, unless otherwise stated.
- 15.9 <u>Independent Contractors</u>. Supplier is an independent contractor, with all of the attendant rights and liabilities of an independent contractor, and not an agent or employee of Recipient. Any provision in this Agreement, or any action by Recipient, that may appear to give Recipient the right to direct or control Supplier in performing under this Agreement means that

Supplier will follow the desires of Recipient in results only.

15.10 Notices. All notices and other communications to be given to any Party hereunder are sufficiently given for all purposes hereunder if in writing and delivered by hand, courier or overnight delivery service or three days after being mailed by certified or registered mail, return receipt requested, with appropriate postage prepaid, or when received in the form of a facsimile or e-mail transmission and will be directed to the physical address, facsimile number or e-mail address set forth below (or at such other address or facsimile number as such Party will designate by like notice):

IF TO AFI:

Ally Financial Inc. 200 Renaissance Center, MC 200-B09-B-11 Detroit, Michigan 48265 Attention: Chief Financial Officer E-Mail Address: james.mackey@ally.com

With copies to:

Ally Financial Inc. 200 Renaissance Center, MC 200-B09-B-11 Detroit, Michigan 48265

Attention: Chief Financial Officer, Global Business Functions

E-Mail Address: marc.scheipe@ally.com

and

Ally Financial Inc. Legal Staff 200 Renaissance Center, MC 200-B09-B-11 Detroit, Michigan 48265 Attention: IT Counsel

E-Mail Address: thomas.lynch@ally.com

IF TO BUYER:

Ocwen Financial Corporation 1661 Worthington Road, Suite 100 West Palm Beach, FL 33409

Attention: Secretary

Telecopy Number: (561) 682-8177 Confirmation Number: (561) 682-8957

[Walter Investment Management Corp.

3000 Bayport Drive, Suite 1100

Tampa, FL 33607 Attn: General Counsel

E-mail: sboyd@walterinvestment.com]

- 15.11 <u>Public Announcements.</u> Except as otherwise required by Law, each of AFI and Buyer will consult with the other and obtain the prior written consent of the other before issuing, or permitting any agent or Affiliate to issue, any press releases or otherwise making, or permitting any agent or Affiliate to make, any public statements with respect to this Agreement or the transactions contemplated hereby; <u>provided, however</u>, that the prior written consent of the other Party is not required hereunder with respect to any press release, public announcement or communication that is substantially similar to a press release, public announcement or communication previously issued with the prior written consent of the other Party.
- 15.12 Amendments and Waivers. This Agreement may not be modified or amended except by an instrument or instruments in writing signed by both Parties. The Parties may, only by an instrument in writing, waive compliance by the other Party with any term or provision of this Agreement on the part of such other Party to be performed or complied with. The waiver by either Party of a breach of any term or provision of this Agreement will not be construed as a waiver of any subsequent breach. Except as otherwise expressly provided herein, no failure to exercise, delay in exercising or single or partial exercise of any right, power or remedy by either Party, and no course of dealing between the Parties, will constitute a waiver of any such right, power or remedy.
- **15.13** <u>Severability</u>. If any provision of this Agreement is held invalid, illegal or unenforceable, the validity, legality or enforceability of the other provisions of this Agreement will not be affected thereby, and there will be deemed substituted for the provision at issue a valid, legal and enforceable provision as similar as possible to the provision at issue.
- 15.14 No Third Party Beneficiaries. This Agreement is entered into solely between, and, except with respect to the rights of the Supplier Indemnified Parties and the Recipient Indemnified Parties under Article 13, and the rights of Buyer's Affiliates and Buyer's and its Affiliates' respective directors, officers, members, managers and employees under Section 15.17(a), may be enforced only by, AFI and Buyer. This Agreement does not create any rights or causes of action in or on behalf of any third parties, including employees, suppliers, customers, Affiliates or Subcontractors of a Party, or create any obligations of a Party to any such third parties, except that either Party is entitled to include as part of its Damages under this Agreement the Damages incurred by its Affiliates or Subcontractors in connection with a breach by the other Party of this Agreement.
- **15.15** Order of Precedence. In the event of a conflict, this Agreement takes precedence over the Schedules.
- 15.16 Non-Solicitation of Personnel. Except as otherwise agreed by the Parties, during the Term and for a period of one (1) year following the termination of the last outstanding Applicable Term, neither Party nor its controlled Affiliates shall, as a result of becoming aware of any employee of the other Party who is connected with the performance of this Agreement, directly or indirectly solicit the employment of or hire (or utilize as an independent contractor) such employee. However, neither the publication of classified advertisements in newspapers, periodicals, Internet bulletin boards, or other publications of general availability or circulation

12-12020-mg Doc 2222 Filed 11/19/12 Entered 11/19/12 10:31:22 Main Document Pg 47 of 63

nor the solicitation, consideration and hiring of persons responding to such advertisements shall be deemed a breach of this **Section 15.16**, unless the advertisement and solicitation is undertaken as a means to circumvent or conceal a violation of this provision.

15.17 Additional Agreements Relating to Certain Properties.

(a) AFI acknowledges and agrees that notwithstanding the fact that (i) EPRE LLC may assign its rights under the Purchase Agreement to Buyer and (ii) GMAC Mortgage, LLC may assign its rights under the Assignment of Leasehold Interest to Buyer, Buyer shall not be deemed to assume EPRE LLC's obligations or GMAC Mortgage, LLC's obligations, respectively, under the Purchase Agreement (including Section 8.3(b), thereof) or the Assignment of Leasehold Interest (including Section 5(b), thereof).

[Signature Page Follows]

12-12020-mg Doc 2222 Filed 11/19/12 Entered 11/19/12 10:31:22 Main Document Pg 48 of 63

IN WITNESS WHEREOF, the Parties their authorized representatives, to be effective	s have caused this Agreement to be executed by as of the Effective Date.		
Ally Financial Inc.	Ocwen Loan Servicing, LLC		
Ву:	Ву:		
Name:	Name:		
Title:	Title:		
[Walter Investment Management Corp.]			
Ву:			
Name:			
Title:			

Schedule A-1 AFI Services⁵

The following Statements of Work are governed by the Agreement.

AFI to Buyer Statements of Work by Functional Service Area

[**NTD: Service periods to be specified for each Service. →]

	Functional Service Area	SOW Name	SOW Number (if applicable)	Service Period
1	Audit	Internal Audit Services	1G	
2	Capital Markets	Capital Markets Services	1C	
3	Communications	Communications and Investor Relations Services	-	
4	Communications	eCommerce Services	1H	
5	Compliance	Global Security Services	10	
6	Compliance	Global Privacy Services	1N	
7	Compliance (includes Consent Order Costs)	Global Compliance Services	1M	
8	Facilities	Facilities Services	1Ka	
9	Finance Tax	Finance Shared Services	1a04	
10	Finance Tax	Tax Services	1a08	
11	Finance Tax	General Accounting Services	1A5	
	Government			
12	Relations	Government Relations	1X	
13	Human Resources	Human Resources Support Services	1Qa	
14	ITG	Global Information Security	1R1	
15	ITG	Application Development and Support – IT Corporate Services	1R9	
16	ITG	Cross Functional Services	-	
17	ITG	End User Computing Services	-	
18	ITG	IT Technology Strategy and Architecture Services	-	
19	ITG	Voice Services Mobility SOW (Mid)_AFI to ResCap	-	
20	ITG	Network Services – Local Area Network	-	
21	ITG	Network Services – Network WAN Network Services WAN (Mid)_AFI to ResCap	-	
22	ITG	Hosting Operations – Hosting, Implementation and Data Center Services	-	
23	ITG	Voice Services - Telecommunications and Contact Center Telecom Call Center AFI to ResCap	-	
24	Legal	Legal Services	1S	

⁵ Note to Draft: Not less than 30 days prior to Closing, this Schedule will be revised to list (i) the transition services as contemplated by SOWs under the SSA that remain required to be performed by AFI as of the Closing Date, and (ii) to the extent mutually agreed upon, any additional services that AFI or Affiliates of AFI provided to ResCap and its Affiliates in the 12-month period prior to the Closing Date that are not specifically identified in the SOWs under the SSA. [Also note that (i) there may be additional services to remove from this list when we execute the agreement and (ii) the structure of including or cross referencing to SOWs is to be discussed.]

25	Loan Review	Loan Review	1T	
26	Marketing	Global Brand & Product Marketing	1I	
27	Risk	Risk Management	1J	
28	Supply Chain	Supply Chain Management Services	1P	
29	Treasury	Treasury Services - Global Funding and Liquidity ("GF&L")	1W1	
30	Treasury	Treasury Services Structured Funding Deal Compliance and Facility Management	1W2	
31	Treasury	Treasury Services, Treasury Operations	1W7	

Schedule A-2

Buyer Services⁶

The following Statements of Work are governed by the Agreement. Buyer to AFI Statements of Work by Functional Service Area

[**NTD: service periods are to be specified for each Service. →]

#	Functional Service Area	SOW Name	SOW Number (if applicable)	Service Period
1	Accounting	Capital Markets Accounting	14L	
2	Accounting	ResCap Accounting Services	-	
3	Accounting	Accounting	14F	
4	Business Risk and Controls and Business Excellence Support	Business Risk and Controls and Business Excellence Support	14.J	
5	Capital Markets	Capital Markets	14B	
6	Capital Markets	Records Services	14I	
0	Capital Markets	Line of Business Compliance	1+1	
7	Compliance	Services	14H	
8	Consumer Lending	Consumer Lending Services	14M	
9	Finance	Mortgage Financial Planning and Analysis	14G	
10	Human Resources	HR Support Services (AFI)	1Qb	
11	Human Resources	HR Support Services (Bank)	14C	
12	ITG	IT Resource Services	-	
13	ITG	Application Support	14A	
14	Legal	Legal Services (AFI)	1S	
15	Legal	Legal Services (Bank)	14E	
16	Masterservicing	Master Servicing for Ally Auto - Bond Administration	-	
17	Masterservicing	Master Servicing for Ally Auto - Bond Modeling	-	
18	Risk	Risk Services Reporting and Special Assets	-	
19	Risk	Risk Management & Data Collection	14D	
20	Risk	Client Repurchase	14N	

⁶ Note to Draft: Not less than 30 days prior to Closing, this Schedule will be revised to list (i) the transition services as contemplated by SOWs under the SSA that remain required to be performed by ResCap as of the Closing Date, and (ii) to the extent mutually agreed upon, any additional services that ResCap or Affiliates of ResCap provided to AFI in the 12-month period prior to the Closing Date that are not specifically identified in the SOWs under the SSA.

Schedule A-3

AFI Provided Migration Assistance Services 7

 $^{^{7}}$ Note to Buyer: The Parties to discuss the Migration Assistance Services that are required by each Party.

Schedule A-4

Buyer Provided Migration Assistance Services⁸

 $^{^{8}}$ Note to Buyer: The Parties to discuss the Migration Assistance Services that are required by each Party.

Schedule B

Governance Model

1. INTRODUCTION

2. AGREEMENT COVERAGE

This <u>Schedule B</u> sets out the service management model for Supplier and Recipient (the "Service Management Model"), related to the Services provided under the Agreement.

3. OVERVIEW

Supplier and Recipient will each appoint responsible, knowledgeable persons to serve as their representatives to oversee the performance of the Agreement (such persons and each Party's Relationship Manager shall collectively constitute the "Service Management Team").

The Service Management Team members will work informally on a regular basis, but the Relationship Managers, and such additional Service Management Team members as the Relationship Managers may designate from time to time, will meet formally at least monthly. These monthly meetings will review such topics as significant operational concerns, transition service progress and pricing. In addition, the Relationship Managers may from time to time designate Service Management Team members to meet in subteams either on a temporary or ongoing basis, to deal with specific issues under the Agreement. The Relationship Managers will share the responsibility for organizing the meetings and ensuring the effectiveness of the meetings.

Each Party will staff its Service Management Team as necessary to represent the areas of services offered.

4. SCOPE

The Service Management Team will be responsible for overseeing the overall execution of the Agreement to ensure that each Party performs its responsibilities as expected.

The Service Management Team's responsibilities shall include the following:

- (a) <u>Strategy and Direction</u>. The Service Management Team may make recommendations as to strategic direction and provide critical input for business planning decisions between Supplier and Recipient.
- (b) General Service Management. The Service Management Team may review the

CONFIDENTIAL
AMECURRENT 703734421.6 30-Oct-12 10:13
AMECURRENT 703734421.10 16-Nov-12 14:56

Schedule B

Page 1

(c) <u>Issue Resolution</u>. The Service Management Team will be responsible for resolving problems, concerns or inefficiencies in their execution as they arise, and will escalate unresolved issues as described below.

5. ESCALATION

Issues that cannot be resolved within a reasonable period of time (which period of time shall not exceed ten (10) days) by the Parties' respective Service Management Team members responsible for the applicable subject area must be first be escalated to the Relationship Managers. If the Relationship Managers are unable to resolve an issue within five (5) days, then either Party's Relationship Manager may, upon prior written notice to the other Relationship Manager, escalate such issue to Recipient's and Supplier's executive management representative for resolution. Neither Party may initiate any formal legal proceedings for resolution of such issue until five (5) business days after such issue has been escalated to each Party's executive management representatives.

The provisions and time periods specified in this <u>Article 5</u> will not be construed to prevent a Party from instituting, and a Party is authorized to institute, formal proceedings earlier to (A) avoid the expiration of any applicable limitations period, (B) preserve a superior position with respect to other creditors, (C) address a claim arising out of the breach of a Party's obligations under <u>Article 9</u> of the Agreement or a dispute with respect to Intellectual Property Rights or (D) seek injunctive relief to prevent or stop a breach or threatened breach of this Agreement.

Schedule C

Pricing Methodology⁹

⁹ Note to Buyer: AFI is continuing to review <u>The Schedule C (pPricing Methodologyfor Services under this Agreement will be based on the cost allocation methodology used to determine the pricing under the SSA, with adjustments as agreed upon by the parties.) and will provide that document to Buyer separately.</u>

Schedule D-1

List of Buyer Recipients and Supported Facilities

[**To be attached]

Schedule D-2

List of AFI Recipients and Supported Facilities

[**To be attached]

Schedule E

Form of Supplement

.....

Supplement [X]

This Supplement __ (this "Supplement") is made and entered into as of the __ day of ___, 2012 (the "Supplement Effective Date") by and [among] Ally Financial Inc., a Delaware corporation ("AFI") and Ocwen Loan Servicing, LLC ("Ocwen" [or "Buyer")] [and Walter Investment Management Corp. ("Walter"); collectively with Ocwen, "Buyer")] (together with AFI, the "Parties").

1. BACKGROUND

This Supplement is entered into pursuant to the terms of the Transition Services Agreement between AFI and [Buyer] dated [•] (the "Agreement") and constitutes a Supplement under the Agreement. Capitalized terms used but not defined in this Supplement have the meanings assigned to those terms in the Agreement.

2. SERVICES DESCRIPTION AND CHARGES

Supplier will provide [Insert brief description of the services here.]____ services as [Additional Services]/[Customized Services][, which shall be described in more detail in ______, for the Charges set forth [therein]/[in Schedule C to the Agreement].

3. CHANGES

[Refer to the clauses in the Agreement relating to Services and incorporate here as appropriate and agreed as the clauses in the Agreement do not apply to the Additional Services or Customized Services being added pursuant to this Supplement.]

4. TERM AND TERMINATION

[Term and notice provisions to be inserted as appropriate.]

5. **[OTHER TERMS**]

The Parties further agree:

[Insert any other terms and conditions applicable to the Additional Services or Customized Services to be performed under this Supplement.]

6. MISCELLANEOUS

This Supplement is incorporated by reference into the Agreement. In the event of any conflict between the terms of this Supplement and the Agreement, the terms

of this Supplement shall only prevail to the extent that this Supplement expressly states that it is intended to override a term of the Agreement.

SPACE BELOW INTENTIONALLY BLANK – SIGNATURE PAGE FOLLOWS

IN WITNESS WHEREOF, the Parties have their respective duly authorized representative		
Ally Financial Inc.	Ocwen Loan Servicing, LLC	
Ву:	By:	
Name:		
Title:	Title:	
[Walter Investment Management Corp.]		
Ву:	_	
Name:	_	
Title:	_	

Schedule F-1

List of Pre-Approved AFI Subcontractors

[<mark>**To be attached</mark>]

Schedule F-1

List of Pre-Approved Buyer Subcontractors

[<mark>**To be attached</mark>]